### Annals of Mathematics

Dehn Surgery on Knots Author(s): Marc Culler, C. McA. Gordon, J. Luecke and Peter B. Shalen Source: Annals of Mathematics, Second Series, Vol. 125, No. 2 (Mar., 1987), pp. 237-300 Published by: Annals of Mathematics Stable URL: <u>http://www.jstor.org/stable/1971311</u> Accessed: 04/01/2015 21:37

Your use of the JSTOR archive indicates your acceptance of the Terms & Conditions of Use, available at http://www.jstor.org/page/info/about/policies/terms.jsp

JSTOR is a not-for-profit service that helps scholars, researchers, and students discover, use, and build upon a wide range of content in a trusted digital archive. We use information technology and tools to increase productivity and facilitate new forms of scholarship. For more information about JSTOR, please contact support@jstor.org.



Annals of Mathematics is collaborating with JSTOR to digitize, preserve and extend access to Annals of Mathematics.

http://www.jstor.org

## Dehn surgery on knots

By Marc Culler<sup>1</sup>, C. McA. Gordon<sup>2</sup>, J. Luecke<sup>3</sup>, and Peter B. Shalen<sup>4</sup>

#### Introduction

We consider the problem of which Dehn surgeries on a knot can produce 3-manifolds with cyclic fundamental group. It is natural to work in the following setting. Let M be a compact, connected, irreducible, orientable 3-manifold such that  $\partial M$  is a torus. The unoriented isotopy class of a non-trivial simple closed curve in  $\partial M$  will be called its *slope*. For any slope r, a closed 3-manifold M(r)may be constructed by attaching a solid torus J to M so that a curve of slope rbounds a disk in J.

If r and s are two slopes, we denote their (minimal) geometric intersection number by  $\Delta(r, s)$ .

The main result of this paper is the following theorem.

CYCLIC SURGERY THEOREM. Suppose that M is not a Seifert fibered space. If  $\pi_1(M(r))$  and  $\pi_1(M(s))$  are cyclic, then  $\Delta(r, s) \leq 1$ . Hence there are at most three slopes r such that  $\pi_1(M(r))$  is cyclic.

This result is sharp; Fintushel-Stern have shown (private communication) that 18- and 19-surgeries on the (-2, 3, 7) pretzel knot yield lens spaces. Many examples of the same phenomenon have been produced by Berge (private communication).

For a detailed account of the manifolds obtained by attaching solid tori to a Seifert fibered space along some of its boundary components, see [Hei].

We give some corollaries which apply to Dehn surgery on knots K in S<sup>3</sup>. In the proofs, M will denote the complement of an open tubular neighborhood of

<sup>&</sup>lt;sup>1</sup>Partially supported by NSF grant NSF-DMS-84-02595.

<sup>&</sup>lt;sup>2</sup>Partially supported by NSF grant NSF-DMS-84-03670.

<sup>&</sup>lt;sup>3</sup>Partially supported by an NSF post-doctoral Fellowship.

<sup>&</sup>lt;sup>4</sup> Partially supported by NSF grant NSF-DMS-86-02433.

K. We use  $\mathbf{Q} \cup \{1/0\}$  as in [R] to parametrize the slopes on  $\partial M$ , and denote M(r) by K(r). In particular,  $K(1/0) = S^3$ .

COROLLARY 1. If K is not a torus knot and  $r \in \mathbf{Q}$ , then  $\pi_1(K(r))$  can be cyclic only if r is an integer. Moreover, there are at most two such integers r, and if there are two then they must be successive.

*Proof.* The manifold M is Seifert-fibered if and only if K is a torus knot. Also,  $\Delta(a/b, 1/0) = |b|$ , and if  $n_1$  and  $n_2$  are integers, then  $\Delta(n_1, n_2) = |n_1 - n_2|$ .

The next corollary is a step towards establishing the conjecture that any nontrivial knot has Property P, in other words, that K(r) is not simply-connected if  $r \in \mathbf{Q}$ .

COROLLARY 2. If K is a non-trivial knot and  $r \in \mathbf{Q}$  is not equal to 1 or -1 then K(r) is not simply-connected. Moreover, K(1) and K(-1) cannot both be simply-connected.

*Proof.* This follows from Corollary 1 and the fact that non-trivial torus knots have Property P [Ms].  $\Box$ 

COROLLARY 3. Up to unoriented equivalence, there are at most two knots whose complements are of a given topological type.

*Proof.* Suppose that there are three inequivalent knots whose complements are homeomorphic. Then for any one of these knots, say K, there exist two distinct rational numbers r and s such that  $K(r) = K(s) = K(1/0) = S^3$ . This contradicts Corollary 2.

COROLLARY 4. If K is a non-trivial amphicheiral knot and  $r \in \mathbf{Q} - \{0\}$ , then  $\pi_1(K(r))$  is not cyclic. In particular, K has Property P.

*Proof.* If K is amphicheiral, then there is an automorphism of M which takes r to -r for all slopes  $r \in \mathbf{Q} \cup \{1/0\}$ . Since  $\Delta(r, -r) \ge 2$  if  $r \in \mathbf{Q} - \{0\}$ , and since torus knots are not amphicheiral, the result follows.  $\Box$ 

COROLLARY 5. Knots of Arf invariant 1 are determined up to unoriented equivalence by their complements.

**Proof.** As in the proof of Corollary 3, it follows from Corollary 2 that if K is not determined by its complement then either K(1) or K(-1) is homeomorphic to S<sup>3</sup>. But if  $\alpha(K) \in \mathbb{Z}_2$  denotes the Arf invariant of K, then the Rohlin invariant of K(1/n) is  $n\alpha(K)$  (see [Gn]).

Actually, a considerably stronger result follows from recent work of Casson's [C], [A-M]. He proves that any knot K whose Alexander polynomial  $\Delta(t)$  satisfies  $\Delta''(1) \neq 0$  has Property P. In particular, since the mod 2 reduction of  $\frac{1}{2}\Delta''(1)$  is  $\alpha(K)$ , this implies that knots of Arf invariant 1 have Property P.

It is shown in [Wh] that Corollary 1 implies the following result.

COROLLARY 6. Prime knots with isomorphic groups have homeomorphic complements.  $\hfill \Box$ 

Finally, combining our work with the fact, recently proved by Bleiler and Scharlemann [B-S], that strongly invertible knots have Property P, we can state the following result.

COROLLARY 7. If K is a non-trivial knot which is invariant under a non-trivial periodic automorphism of  $S^3$ , then K has Property P.

This will be proved in Section 2.8.

Before giving some indications of the method of proof of the Cyclic Surgery Theorem, we shall fix some conventions and terminology that will be used throughout the paper. We will work in the smooth category. All manifolds are understood to be orientable. If S is a surface in a 3-manifold, we define a *compressing disk* for S to be a disk  $D \subset M$  such that  $D \cap S = \partial D$  and such that  $\partial D$  is (homotopically) non-trivial in S. If S is properly embedded one can surger S using D to obtain a new properly embedded surface in M; this operation will be called a *compression*. A surface S in a 3-manifold M will be called *incompressible* if (i) no component of S is a sphere and (ii) there is no compressing disk for S in M. The incompressible surfaces that we discuss will all be either properly embedded (always the case in Chapter 1) or contained in the boundary of the ambient 3-manifold (sometimes the case in Chapter 2). In these cases, (ii) is equivalent to the condition that the fundamental group of each component of S maps injectively to that of M.

By an essential surface in a 3-manifold M, we shall mean a properly embedded surface which is incompressible and no component of which is parallel to a subsurface of  $\partial M$ . Now suppose that M is irreducible and that  $\partial M$ is a torus, and let S be any essential surface in M with  $\partial S$  non-empty. The boundary components of S are disjoint non-trivial simple closed curves in  $\partial M$ and hence all have the same slope, say r. We call r the boundary slope of S. A slope r on  $\partial M$  will be called a boundary slope if it is the boundary slope of some essential surface in M. We will call r a strict boundary slope if it is the boundary slope of some essential surface which is not a fiber in any fibration of M over the circle.

The Cyclic Surgery Theorem is proved in Chapter 1 for the case when Mcontains no essential torus and neither of the given slopes r, s is a strict boundary slope. Here the argument is based on the Thurston Geometrization Theorem, which asserts in this situation that the interior of M has a hyperbolic metric of finite volume. As in [C-S1], this is used to define a family of (characters of) representations of  $\pi_1(M)$  in SL<sub>2</sub>(C), parametrized by the points of a complex affine algebraic curve  $X_0$ . The ideal points of a de-singularized projective completion  $\tilde{X}_0$  of  $X_0$  correspond to actions of  $\pi_1(M)$  on trees, which can be used to define essential surfaces in M. One shows that for all but a very small set \* of slopes r, one of the following alternatives holds. Either there is a point of  $X_0$  corresponding to a representation of  $\pi_1(M)$  in  $SL_2(C)$  which induces a representation of the quotient group  $\pi_1(M(r))$  onto a non-cyclic subgroup of  $PSL_{2}(C)$ ; or there is some ideal point of  $\tilde{X}_{0}$  which defines an essential surface S which is closed or has boundary slope r, and is not a fiber. Furthermore, if S is closed then it has positive genus and remains incompressible in M(r). Thus in each of these cases, either  $\pi_1(M(r))$  is non-cyclic or r is a strict boundary slope.

For a more precise outline of the proof in this case, see Sections 1.0 and 1.1.

Chapter 2 is devoted to the proof of the Cyclic Surgery Theorem in the case where either M contains an essential torus or one of the given slopes is a strict boundary slope. One shows that it is enough to prove the theorem for manifolds which are not cabled (in the sense of [Gr-L]). If M is not cabled and contains an essential torus T, it is shown that T remains incompressible in M(r) for all but a very small set of slopes r. This is done by considering two slopes r and s such that T compresses in M(r) and M(s), and carrying out a graph-theoretic analysis of the intersection of the two planar surfaces in M corresponding to the compressing disks for T in M(r) and M(s). An analogous argument allows one to deal also with the case where the first Betti number of M is greater than 1. There, by finding restrictions on the slopes r such that M(r) is reducible, it is shown that  $\pi_1(M(r))$  is non-cyclic for all but a very small set of slopes r. For the case of boundary slopes, observe that if r is the boundary slope of an essential surface F in M, then F can be capped off to produce a closed surface  $\hat{F}$  in M(r). Assuming (as we now may) that the first Betti number of M is 1 and that r is a strict boundary slope, it is shown that if F is suitably chosen then either  $\hat{F}$  is incompressible in M(r), or  $\hat{F}$  is a 2-sphere which decomposes M(r) as a connected sum of two non-trivial lens spaces, or M contains a closed essential surface S with special properties. Finally, by a graph-theoretic analysis of the

240

<sup>\*</sup>For the purpose of this informal discussion, a set of slopes is "very small" if  $\Delta(r, s) \leq 1$  for all r and s in the set.

kind mentioned above, one shows that S remains incompressible in M(s) whenever  $\Delta(r, s) > 1$ .

A more detailed summary of the contents of Chapter 2 is given in Section 2.0.

The proof of the Cyclic Surgery Theorem gives a rather stronger result. Let us define a closed 3-manifold L to be *small* if

(\*) there exists no incompressible surface in L; and

(\*\*) there exists no representation of  $\pi_1(L)$  into  $PSL_2(\mathbb{C})$  with non-cyclic image.

Then in both the statement and proof of the Cyclic Surgery Theorem, the hypothesis that M(r) and M(s) have cyclic fundamental groups may be replaced by the condition that they are small. (A connected sum of two non-trivial lens spaces violates (\*\*) because a free product of two cyclic groups is Fuchsian and hence embeds in  $PSL_2(\mathbf{R})$ .)

We thank F. Bonahon for some helpful insights regarding this work. We also thank E. Luft and R. A. Litherland, for pointing out an error in one of the arguments in the first version of this paper, and G. Mess and R. Skora for supplying a number of corrections to that manuscript.

#### Chapter 1

In this chapter we will prove the following case of the Cyclic Surgery Theorem.

THEOREM 1.0.1. Let M be a compact irreducible (orientable) 3-manifold with torus boundary. Suppose that M is not a Seifert fibered space and contains no essential torus. Let r and s be slopes such that  $\pi_1(M(r))$  and  $\pi_1(M(s))$  are cyclic groups. If neither r nor s is a strict boundary slope then  $\Delta(r, s) \leq 1$ .

In Section 1.1 we will reduce the proof of this theorem to that of two other results, Propositions 1.1.2 and 1.1.3. Proposition 1.1.2 will be proved in Section 1.4, and Proposition 1.1.3 will be proved in Sections 1.5 and 1.6. In Sections 1.2 and 1.3 we reformulate and extend some of the ideas in [C-S1] to make them more directly applicable to the proofs of these propositions.

Throughout this chapter it is understood that M satisfies the hypotheses of Theorem 1.0.1. Thus it follows from the Thurston Geometrization Theorem [MB, Chapter V] that the interior of M admits a hyperbolic metric of finite volume.

1.0.2. We need to establish some conventions regarding the use of fundamental groups in order to avoid problems with base points. If we are given a connected polyhedron X and if we have fixed a universal cover  $\tilde{X}$  of X, then we will denote by  $\pi_1(X)$  the group of covering transformations of  $\tilde{X}$ . If x is a point of X then a lift  $\tilde{x} \in \tilde{X}$  of x determines an identification of  $\pi_1(X)$  with  $\pi_1(X, x)$ . An oriented closed curve in X determines, up to conjugacy, an element of  $\pi_1(X)$ . A map from a connected polyhedron Y to X determines, up to composition with inner automorphisms, a homomorphism from  $\pi_1(Y)$  to  $\pi_1(X)$ . Thus many of the statements which are proved in this chapter apply to objects which are only defined up to conjugacy. Our convention is that this ambiguity will be ignored provided that the statements in question are invariant under conjugacy.

We fix, for the entire chapter, a universal cover  $\tilde{M}$  of M.

It will be convenient to adopt the following notation. We will denote by L the group  $H_1(\partial M; \mathbb{Z})$ , which will be regarded as a lattice in the 2-dimensional real vector space  $V = H_1(\partial M; \mathbb{R})$ . There is a homomorphism  $\pi_1(\partial M) \to \pi_1(M)$  which is defined up to composition with inner automorphisms of  $\pi_1(M)$ . We will identify  $\pi_1(\partial M)$  with its image under this homomorphism. We will let  $e: L \to \pi_1(\partial M)$  denote the inverse of the Hurewicz isomorphism. We will often view e as a homomorphism from L to  $\pi_1(M)$  which is defined modulo inner automorphisms. Recall that a slope is an isotopy class of unoriented simple closed curves on  $\partial M$ . Thus each slope r corresponds to a pair  $\{\pm \alpha\}$  of primitive elements of L; we shall write  $M(\alpha) = M(r)$ . Note that  $\pi_1(M(\alpha))$  has presentation  $|\pi_1(M): e(\alpha) = 1|$ . If r is a (strict) boundary slope then we will call  $\alpha$  and  $-\alpha$  (strict) boundary classes. If  $\alpha$  and  $\beta$  are elements of L then we will write  $\Delta(\alpha, \beta)$  for the absolute value of their intersection number.

#### 1.1. The curve of characters

We will briefly review some notation and basic theorems from [C-S1] and [C-S2]. We denote by  $R = R(\pi_1(M))$  the space of all representations of  $\pi_1(M)$  in  $SL_2(C)$ . This space has the structure of a complex affine algebraic set. The set  $X = X(\pi_1(M))$  is the set of characters of representations in R, and  $t: R \to X$  is the natural map which sends a representation  $\rho$  to its character  $\chi_{\rho}$ . This set also has the structure of a complex affine algebraic set [C-S1, Corollary 1.4.5]. The map t is a regular map, and for each  $\gamma \in \pi_1(M)$  there is a regular function  $I_{\gamma}: X \to C$  defined by  $I_{\gamma}(\chi) = \chi(\gamma)$ . The set of functions  $I_{\gamma}$  for  $\gamma \in \pi_1(M)$  generates the coordinate ring of X. Note that since  $I_{\gamma}$  depends only upon the conjugacy class of  $\gamma \in \pi_1(M)$ ,  $I_{e(\alpha)}$  is a well-defined function on X for each  $\alpha \in L$ . Since M is hyperbolic there is a representation  $\rho_0 \in R$  which is discrete and faithful [C-S1, Proposition 3.1.1]. We denote by  $R_0$  an irreducible component of R containing  $\rho_0$ , and set  $X_0 = t(R_0)$ .

The following proposition is essentially due to Thurston.

PROPOSITION 1.1.1. The set  $X_0 \subset X$  is an irreducible affine variety of dimension 1. For each non-trivial element  $\alpha$  of L, the function  $I_{e(\alpha)}$  is non-constant on  $X_0$ .

*Proof.* Proposition 1.4.4 of [C-S1] shows that  $X_0$  is a variety. By Proposition 3.2.1 of [C-S1], dim  $X_0 \ge 1$ . The equality dim  $X_0 = 1$  then follows from Proposition 2 of [C-S2], which also shows that  $I_{e(\alpha)}$  is non-constant on  $X_0$  for each non-trivial element  $\alpha$  of L.

The idea of the proof of Theorem 1.0.1 is to show, for most  $\alpha \in L$ , that  $\pi_1(M(\alpha))$  is non-cyclic by producing a representation in  $\text{PSL}_2(\mathbb{C})$  with non-cyclic image. A representation  $\rho \in R_0$  will induce such a representation if and only if  $\rho(e(\alpha)) = \pm 1$ . An obvious necessary condition for  $\rho(e(\alpha)) = \pm 1$  is that the value of the function  $I_{e(\alpha)}$  should be  $\pm 2$  at the point  $t(\rho) \in X_0$ . This suggests studying the zeroes of the functions  $f_{\alpha}: X_0 \to \mathbb{C}$  defined for all  $\alpha \in L$  by

$$f_{\alpha} = I_{e(\alpha)}^2 - 4.$$

It is always easier to study the zeroes of a function defined on a smooth projective curve. Thus, as in [C-S1], we will denote by  $\tilde{X}_0$  the smooth projective curve which is birationally equivalent to  $X_0$ . The birational equivalence from  $\tilde{X}_0$  to  $X_0$  is regular (i.e. defined) at all but a finite number of points of  $\tilde{X}_0$ . The points where it is defined will be called *ordinary points*, and the others will be called *ideal points*. We will identify the function fields of  $X_0$  and  $\tilde{X}_0$  under the isomorphism induced by the birational equivalence. Thus any rational function f on  $X_0$  pulls back to a rational function on  $\tilde{X}_0$  which will also be denoted f.

Recall that a rational function defined on a smooth curve is meromorphic and that its degree is equal to the number of poles, counted with multiplicities. For a non-zero rational function the degree is also equal to the number of zeroes. The following proposition describes the behavior of the degree of  $f_{\alpha}$ :  $X_0 \to \mathbf{C} \cup \{\infty\}$  as  $\alpha$  varies over the group L.

**PROPOSITION 1.1.2.** There exists a norm  $\|\cdot\|$  on the real vector space  $V = H_1(\partial M; \mathbf{R})$  with the following properties.

(i) For each  $\alpha \in L$ ,  $\|\alpha\| = \text{degree } f_{\alpha}$ .

(ii) The unit ball is a finite-sided polygon whose vertices are rational multiples of strict boundary classes in L.

This will be proved in Section 1.4 by examination of the poles of the functions  $f_{\alpha}$  for  $\alpha \in L$ .

The next proposition concerns the orders of zeroes of the functions  $f_{\alpha}$  for  $\alpha \in L$ . If x is a point of  $\tilde{X}_0$  and f is a rational function on  $\tilde{X}_0$  with f(x) = 0, then we will denote by  $Z_x(f)$  the order of zero of f at x. If  $f(x) \neq 0$ , and in particular if f has a pole at x, we set  $Z_x(f) = 0$ .

**PROPOSITION 1.1.3.** Suppose that  $\alpha$  is a primitive element of L which is not a strict boundary class, and that  $\pi_1(M(\alpha))$  is cyclic. Then, for each point  $x \in \tilde{X}_0$ , and for each non-zero  $\delta \in L$ , we have

$$Z_{\mathbf{x}}(f_{\alpha}) \leq Z_{\mathbf{x}}(f_{\delta}).$$

This will be proved in Sections 1.5 and 1.6. It will be shown in Section 1.5 that if x is an ordinary point and  $\alpha$  and  $\delta$  are non-zero elements of L such that  $Z_x(f_{\alpha}) > Z_x(f_{\delta})$ , then the image of x in  $X_0$  is the character of a representation  $\rho \in R_0$  which induces a homomorphism of  $\pi_1(M(\alpha))$  onto a non-cyclic subgroup of PSL<sub>2</sub>(C). In the case where x is an ideal point and  $\alpha$  and  $\delta$  are non-zero elements of L such that  $Z_x(f_{\alpha}) > Z_x(f_{\delta})$ , it will be shown in Section 1.6 that either  $\alpha$  is a strict boundary class or there exists a closed surface in M which is incompressible in  $M(\alpha)$ .

The rest of this section will be devoted to the proof that Propositions 1.1.2 and 1.1.3 imply Theorem 1.0.1.

Note that the following is a corollary to Propositions 1.1.2 and 1.1.3.

COROLLARY 1.1.4. Suppose that  $\alpha$  is a primitive element of L which is not a strict boundary class, and that  $\pi_1(M(\alpha))$  is cyclic. Then for each non-zero  $\delta \in L$   $||\alpha|| \leq ||\delta||$ .

Define  $m = \min_{0 \neq \delta \in L} ||\delta||$ , and consider the ball *B* of radius *m* in *V*. By Proposition 1.1.2 this is a compact, convex, finite-sided polygon which is balanced (i.e. -B = B). By construction there are no non-zero elements of *L* contained in the interior of *B*.

We will use the obvious area element defined on V in which any pair of generators of L spans a parallelogram of area 1. The above properties of B imply that the projection of V to the torus V/2L is one-to-one on the interior of B. Thus the area of B is at most 4. (This observation is due to Minkowski.)

Corollary 1.1.4 implies that if  $\alpha \in L$  is a not a strict boundary class and  $\pi_1(M(\alpha))$  is cyclic then  $\alpha$  must lie on the boundary of *B*. Now let  $\alpha$  and  $\beta$  be two elements of *L*, neither of which is a strict boundary class. Suppose that  $\pi_1(M(\alpha))$  and  $\pi_1(M(\beta))$  are cyclic. Let *P* denote the parallelogram in *V* with vertices  $\pm \alpha, \pm \beta$ . We have

$$\Delta(\alpha, \beta) = \frac{1}{2}$$
Area  $P \leq \frac{1}{2}$  Area  $B \leq 2$ .

245

Furthermore, if  $\Delta(\alpha, \beta) = 2$  then we must have B = P. Hence in this case,  $\alpha$  and  $\beta$  are vertices of B. But by Proposition 1.1.2(ii), this implies that they are strict boundary classes, a contradiction. Thus we conclude that  $\Delta(\alpha, \beta) \leq 1$ .

This completes the proof that Propositions 1.1.2 and 1.1.3 imply Theorem 1.0.1.  $\hfill \Box$ 

#### 1.2. Ideal points and trees

In this section we redo the material from Sections 2 and 3 of [C-S1] in a form, suggested by the approach of [M-S1], which is better adapted for the applications in this paper. We begin by reviewing some of the ideas in [Se].

Let F be a field and  $v: F^* \to \mathbb{Z}$  a (discrete) valuation. We denote by  $\mathcal{O}$  the valuation ring  $\{f \in F | f = 0 \text{ or } v(f) \ge 0\}$ , and by  $\mathscr{M}$  the maximal ideal in  $\mathcal{O}$ . The group  $\mathrm{SL}_2(F)$  acts on a tree  $T_F$  which is constructed as follows. Define a *lattice* to be a finitely generated  $\mathcal{O}$ -submodule of the vector space  $F^2$  which spans over F. Define an equivalence relation on lattices by  $\Lambda \sim \Lambda'$  if  $\Lambda = \alpha \Lambda'$  for some  $\alpha \in F^*$ . Given lattices  $\Lambda$  and  $\Lambda'$  there exists an element  $\alpha \in F^*$  such that  $\alpha \Lambda' \subset \Lambda$  and  $\Lambda/\alpha \Lambda'$  is a cyclic  $\mathcal{O}$ -module. The annihilator of this cyclic module is determined by the equivalence classes  $[\Lambda]$  and  $[\Lambda']$ , and is equal to  $\mathscr{M}^d$  for some integer  $d = d([\Lambda], [\Lambda'])$ .

Let T denote the set of equivalence classes of lattices. The function  $d: T \times T \to T$  is an integer-valued metric. Consider the 1-complex constructed by taking T as the vertex set and adding an edge joining each pair of vertices which are distance 1 apart. It is shown in [Se] that this is a tree; i.e. it is connected and simply-connected. For two vertices s and s', the number of edges in the shortest path joining them is given by d(s, s'). We shall denote this 1-complex by T as well.

1.2.1. The obvious action of  $\operatorname{GL}_2(F)$  on the set of lattices in  $F^2$  determines an action on T, and this action is transitive on vertices. The lattice  $\mathcal{O}^2$  generated by the standard basis of  $F^2$  is stabilized by the subgroup  $\operatorname{GL}_2(\mathcal{O})$  of  $\operatorname{GL}_2(F)$ . This subgroup is, in fact, the entire stabilizer of the vertex  $[\mathcal{O}^2]$ . Restricting the action of  $\operatorname{GL}_2(F)$  yields an action of  $\operatorname{SL}_2(F)$  on T, with no inversions, such that the stabilizer of a vertex is a conjugate in  $\operatorname{GL}_2(F)$  of the subgroup  $\operatorname{SL}_2(\mathcal{O})$ .

Next we consider the action of  $SL_2(\mathcal{O})$  on the link  $\mathscr{L}$  of the standard vertex  $[\mathcal{O}^2]$ . (Of course, this action is conjugate to the action of the stabilizer of any vertex on the link of the vertex.)

1.2.2. The vertices of  $\mathscr{L}$  are represented by lattices  $\Lambda \subset \mathscr{O}^2$  such that the quotient  $\mathscr{O}^2/\Lambda$  is isomorphic as an  $\mathscr{O}$ -module to the residue field  $k = \mathscr{O}/\mathscr{M}$ . The image of such a lattice in the 2-dimensional k-vector space  $\mathscr{O}^2/\mathscr{M}^2 = k^2$  is a

1-dimensional subspace. This establishes a bijection between  $\mathscr{L}$  and the projective line  $P^{1}(k)$ ; we will use this to identify  $\mathscr{L}$  with  $P^{1}(k)$ . Then the action of  $SL_{2}(\mathcal{O})$  on  $\mathscr{L}$  is the pull-back, under the obvious homomorphism  $SL_{2}(\mathcal{O}) \rightarrow PSL_{2}(k)$ , of the natural action of  $PSL_{2}(k)$  on  $P^{1}(k)$ . In particular, an element of  $SL_{2}(\mathcal{O})$  acts trivially on  $\mathscr{L}$  if and only if it is congruent to  $\pm 1$  modulo  $\mathscr{M}$ .

In some of the arguments below we will use the following version of the extension theorem for valuations.

THEOREM 1.2.3. Let F be a finitely generated extension of a field K. Let  $w: K^* \to \mathbb{Z}$  be a valuation. Then there exists a valuation  $v: F^* \to \mathbb{Z}$  such that  $v|K^* = d \cdot w$  for some positive integer d.

This is slightly more general than the usual form [Jb] of the extension theorem since F is allowed to be a transcendental extension of K. For a proof see, for example, [M-S1, Lemma II.4.4]

If v and w are as in the above theorem we will say that v is an *extension* of w.

We will now consider a 3-manifold M as in the introduction to the chapter, and the associated varieties  $R_0$ ,  $X_0$  and  $\tilde{X}_0$  defined in Section 1.1. Let K be the function field of  $X_0$ , which we have identified with that of  $\tilde{X}_0$ , and let F be the function field of  $R_0$ . Since there is a regular map of  $R_0$  onto  $X_0$ , we may regard F as an extension field of K.

1.2.4. Recall, from Section 1 of [C-S1], that there is a tautological representation  $P: \pi_1(M) \to SL_2(F)$  defined by

$$P(\gamma) = \begin{bmatrix} a & b \\ c & d \end{bmatrix}$$

where the functions a, b, c, and d are defined by

$$ho(\gamma) = egin{bmatrix} a(
ho) & b(
ho) \ c(
ho) & d(
ho) \end{bmatrix}$$

for all  $\rho \in R_0$ . Note that for any element  $\gamma$  of  $\pi_1(M)$ , the trace of  $P(\gamma)$  is equal to  $I_{\gamma} \in K \subset F$ . Note that since  $\rho_0$  is faithful, P is also faithful.

1.2.5. An ideal point x of  $\tilde{X}_0$  determines a valuation  $w: K^* \to \mathbb{Z}$ , which can therefore be extended to a valuation  $v: F^* \to \mathbb{Z}$ . This valuation determines an action of  $SL_2(F)$  on a tree T which pulls back, under the homomorphism P, to an action of  $\pi_1(M)$  on T. An action constructed in this way will be said to be *associated to* the ideal point x.

(The construction of this action depends on the choice of the extended valuation v. However, the action defines a translation length function which is

shown in [M-S1] to be determined by the ideal point x. In most cases the translation length function essentially determines the action; see [C-M]. We will not use these facts here.)

The following proposition is proved, in a slightly different setting, in [C-S1]; we indicate a proof from the point of view of the present paper.

PROPOSITION 1.2.6. Let  $\pi_1(M) \times T \to T$  be an action associated to an ideal point x. An element  $\gamma$  of  $\pi_1(M)$  has a fixed point in T if and only if  $I_{\gamma}(x) \neq \infty$ . No point of T is fixed by the entire group  $\pi_1(M)$ .

*Proof.* Let w and v be as in 1.2.5, and set  $\mathcal{O} = \mathcal{O}_v$ . By 1.2.1,  $\gamma$  has a fixed point in T if and only if  $P(\gamma)$  belongs to a conjugate of  $PSL_2(\mathcal{O})$  in  $PSL_2(F)$ . Using the rational canonical form (cf. [C-S1, Theorem 2.2.1]) one sees that this is equivalent to the condition  $I_{\gamma} = \text{trace } P(\gamma) \in \mathcal{O}$ , which is in turn equivalent to  $I_{\gamma} \in \mathcal{O}_w$ , i.e.  $I_{\gamma}(x) \neq \infty$ . Since x is an ideal point, and since the functions  $I_{\gamma}$  for  $\gamma \in \pi_1(M)$  generate the coordinate ring of  $X_0$ ,  $\mathcal{O}$  cannot contain all of the  $I_{\gamma}$ . Thus no vertex of T is fixed by the entire group  $\pi_1(M)$ .

The final conclusion in the statement of Proposition 1.2.6 can be strengthened as follows.

PROPOSITION 1.2.7. Let  $\pi_1(M) \times T \to T$  be an action associated to an ideal point. Then no non-trivial normal subgroup of  $\pi_1(M)$  fixes a point of T.

The proof of Proposition 1.2.7 depends on the following lemma.

LEMMA 1.2.8. Let F be a field with valuation v and let  $\Gamma$  be a subgroup of  $SL_2(F)$ . Suppose that  $\Gamma$  contains a normal subgroup N, which is not contained in the center of  $SL_2(F)$ , but which is contained in a conjugate of  $SL_2(\mathcal{O})$ . Then either the trace of every element of  $\Gamma$  lies in  $\mathcal{O}_v$  or else  $\Gamma$  contains an abelian subgroup of index at most 2.

*Proof.* Assume that there exists  $\gamma_0 \in \Gamma$  whose trace does not belong to  $\mathcal{O}_v$ . In particular,  $\operatorname{tr}(\gamma_0) \neq \pm 2$ . Thus, after possibly replacing F by a degree 2 extension and extending the valuation, we may assume that  $\gamma_0$  is diagonalizable over F.

Consider the action of  $\Gamma$  on the tree  $T = T_F$ . The fixed set  $T^N$  of the subgroup N is non-empty since N is contained in a conjugate of  $SL_2(\mathcal{O})$ . By [Se, p. 58],  $T^N$  is a subtree of T. Since N is normal,  $T^N$  is invariant under  $\Gamma$ .

Now let  $\gamma$  be any conjugate of  $\gamma_0$ . Since  $\gamma$  does not fix any vertex of T it follows from [Se, p. 63, Prop. 24] that  $\gamma$  has a unique *axis*  $A_{\gamma}$ , i.e. a subcomplex of T which is homeomorphic to **R** and invariant under  $\gamma$ . Moreover, any subtree of T which is invariant under  $\gamma$  must contain  $A_{\gamma}$ . In particular,  $A_{\gamma} \subset T^N$ .

Since  $\gamma$  is diagonalizable in  $SL_2(F)$  the axis  $A_{\gamma}$  can be described explicitly as follows. Consider a basis  $\{e, f\}$  of eigenvectors of  $\gamma$ . For each integer n let  $s_n$ be the vertex of T corresponding to the lattice with basis  $\{\pi^n e, f\}$ , where  $\pi$  is a generator of the maximal ideal in  $\mathcal{O}$ . One checks directly that  $s_n$  is joined to  $s_{n+1}$ by an edge  $e_n$  and that the union of the  $e_n$  and the  $s_n$  is a line in T which is invariant under the action of  $\gamma$ . Thus this line is the axis  $A_{\gamma}$ .

We claim that each element  $\nu$  of N is diagonal in the basis  $\{e, f\}$ . In proving this we may assume that  $\{e, f\}$  is the standard basis of  $F^2$ . We have seen that  $\nu$  fixes all of the vertices of  $A_{\gamma}$ . Let  $\Pi$  denote the matrix  $\begin{bmatrix} \pi & 0 \\ 0 & 1 \end{bmatrix} \in \operatorname{GL}_2(F)$ , which takes  $s_0$  to  $s_n$ . Thus if  $\nu$  equals  $\begin{bmatrix} a & b \\ c & d \end{bmatrix}$ , then for each  $n \in \mathbb{Z}$ , we have that  $\Pi^{-n}\nu\Pi^n = \begin{bmatrix} a & \pi^{-n}b \\ \pi^n c & d \end{bmatrix}$  fixes  $s_0$  and therefore belongs to  $\operatorname{SL}_2(\mathcal{O})$  by 1.2.1. Therefore b = c = 0 so that  $\nu$  is diagonal, proving the claim.

Since N is not contained in the center of  $SL_2(\mathcal{O})$  there is at most one basis of  $F^2$  in which N is diagonal. Therefore all of the conjugates of  $\gamma_0$  are simultaneously diagonalizable. Therefore there exist exactly two lines in  $F^2$ which are invariant under every conjugate of  $\gamma_0$ . Every element of  $\Gamma$  either keeps these lines invariant or interchanges them. It follows that  $\Gamma$  has an abelian subgroup of index at most 2.

Proof of Proposition 1.2.7. Let  $\Gamma = P(\pi_1(M)) \subset SL_2(F)$ , where P is the tautological representation and F is the function field of  $R_0$ . Let v and w be as in 1.2.4. Suppose that there is a non-trivial normal subgroup of  $\pi_1(M)$  which fixes a vertex of T; let N denote its image under P. Since P is faithful and  $\pi_1(M)$  is torsion-free, N cannot be contained in the center of  $SL_2(F)$ . By 1.2.1, N is contained in a conjugate of  $SL_2(\mathcal{O})$ .

By Lemma 1.2.7, either every element of  $P(\pi_1(M))$  has trace in  $\mathcal{O}_v$  or  $\Gamma$  has an abelian subgroup of index at most 2. The first possibility implies that  $I_{\gamma}(x) \neq \infty$  for all  $\gamma$  in  $\Gamma$ . This is absurd since x is an ideal point. The second possibility implies, since P is faithful, that  $\pi_1(M)$  has an abelian subgroup of index at most 2, contradicting the fact that the interior of M is a hyperbolic manifold of finite volume.  $\Box$ 

#### 1.3. Trees and incompressible surfaces

The results of this section are very similar to those of [C-S1, Section 2]. However, for the applications in Section 1.6 of this chapter it is necessary to state and prove them from a rather different point of view, suggested by [M-S1] and [M-S2]. We continue to work with a manifold M as in the introduction to this chapter. Let  $\tilde{M}$  denote the universal cover of M, and let  $p: \tilde{M} \to M$  be the covering projection. Throughout this section we will suppose that we are given an action without inversions of  $\pi_1(M)$  on a (simplicial) tree T. Let E denote the set of midpoints of edges in T.

Definition 1.3.1. A properly embedded surface s in M is associated to the action if there exists a  $\pi_1(M)$ -equivariant map  $\tilde{\phi}: \tilde{M} \to T$  which is transverse to E, and such that  $\tilde{\phi}^{-1}(E) = p^{-1}(S)$ .

PROPOSITION 1.3.2. If S is a surface associated to the action then, for each component D of M - S, the image of  $\pi_1(D)$  in  $\pi_1(M)$  is contained in the stabilizer of a vertex. Thus if there is no fixed point for the action then any associated surface is non-empty.

(Note that the above statement is consistent with the conventions of 1.0.2, since the set of vertex stabilizers in  $\pi_1(M)$  is invariant under conjugation. The same comment applies to Proposition 1.3.4 below.)

*Proof.* The image of  $\pi_1(D)$  in  $\pi_1(M)$  is the stabilizer of a component of  $p^{-1}(D)$ . The equivariance implies that it fixes a vertex of T. In particular if  $S = \emptyset$  then  $\pi_1(M)$  fixes a vertex of T.

In studying surfaces associated to the action of  $\pi_1(M)$  on T it is convenient to work with an alternate characterization. This will use an explicit form of the construction in Section 2 of [C-S1].

1.3.3. As in [Sco-W] we consider the aspherical complex  $\mathscr{K}$  which is the quotient of  $\tilde{M} \times T$  under the diagonal action of  $\pi_1(M)$ . This action realizes  $\tilde{M} \times T$  as the universal cover of  $\mathscr{K}$ . Thus there is a natural isomorphism between  $\pi_1(M)$  and  $\pi_1(\mathscr{K})$ . The action of  $\pi_1(M)$  on T pushes forward to an action of  $\pi_1(\mathscr{K})$  on T. If x is an edge or vertex of T then the stabilizer of x in  $\pi_1(\mathscr{K})$  will be denoted  $\pi_x$ . If e is an edge of T with midpoint m, we let  $\mathscr{K}_e$  denote the quotient of  $\tilde{M} \times m$  under  $\pi_e$ . If s is a vertex of T,  $\Sigma_s$  will denote the closed star of s in the barycentric subdivision of T. The quotient of  $M \times \Sigma_s$  under  $\pi_s$  will be called  $\mathscr{K}_s$ . Notice that the  $\mathscr{K}_e$  are the components of the bicollared subcomplex  $\mathscr{E} = \tilde{M} \times E/\pi_1(\mathscr{K})$  of  $\mathscr{K}$ . The complexes  $\mathscr{K}_s$  are the closures of the components of  $\mathscr{K} - \mathscr{E}$ .

In terms of the complex  $\mathscr{K}$  we can characterize the surfaces associated to actions as follows.

PROPOSITION 1.3.4. A surface S in M is associated to the action of  $\pi_1(M)$  on T if and only if there exists a map  $\phi: M \to \mathcal{K}$ , such that

(i)  $\phi$  induces the natural isomorphism between  $\pi_1(M)$  and  $\pi_1(\mathscr{K})$  up to conjugacy;

(ii)  $\phi$  is transverse to  $\mathscr{E}$ ; and

(iii)  $S = \phi^{-1}(\mathscr{E}).$ 

*Proof.* Given  $\tilde{\phi}$  as in 1.3.1, we may take  $\phi$  to be induced by the equivariant map  $\tilde{\phi} \times \text{id}$  from  $\tilde{M} \times T$  to T. Given  $\phi$  as in the statement of the lemma,  $\tilde{\phi}$  may be constructed by lifting  $\phi$  to a map  $\tilde{M} \to \tilde{M} \times T$  and composing with the projection to T.

1.3.5. We record here some remarks about the complex  $\mathscr{K}$  that will be needed in Section 1.6. Suppose that r is a point of  $\mathscr{E}$ , and that  $\tilde{r}$  is a lift of r to the universal cover  $\tilde{M} \times T$  of  $\mathscr{K}$ . Then  $\tilde{r} \in \tilde{M} \times e$  for a unique edge e. The lift  $\tilde{r}$  determines an identification of  $\pi_1(\mathscr{K}, r)$  with  $\pi_1(\mathscr{K})$ , which is the group of covering transformations of  $\tilde{M} \times T$ . Under this identification we have  $\pi_1(\mathscr{K}_e, r) = \pi_e$  and  $\pi_1(\mathscr{K}_s, r) = \pi_s$ , where s is either vertex adjacent to e.

Let S be a surface in M. By a *trivial region* for S we mean either a closed 3-ball bounded by a component of S or a closed region of parallelism between a component of S and a submanifold of  $\partial M$ .

PROPOSITION 1.3.6. Let S be a surface associated to the action of  $\pi_1(M)$  on T. Suppose that S' is a surface obtained from S either by compressing S along a disk or by deleting the intersection of S with a trivial region for S. Then S' is also associated to the action.

*Proof.* Let S be defined by a map  $\phi: M \to \mathscr{K}$  as in Proposition 1.3.4. Suppose that S' is obtained by compressing S along a disk. Then by performing surgery on the map  $\phi$  (cf. [Hem, proof of Lemma 6.5]), one obtains a homotopic map  $\phi'$  which is transverse to  $\mathscr{E}$  and such that  $(\phi')^{-1}(\mathscr{E}) = S'$ . The conclusion then follows from Proposition 1.3.4.

Next suppose that S' is obtained by deleting the intersection of S with a trivial region R for S. There is a map  $\phi': M \to \mathcal{K}$  which agrees with  $\phi$  outside a small neighborhood V of R and maps V into  $\mathcal{K} - \mathcal{E}$ . Clearly S' is the pre-image of  $\mathcal{E}$  under  $\phi'$ , and  $\phi'$  is homotopic to  $\phi$ . The conclusion again follows from 1.3.4.

COROLLARY 1.3.7. Suppose that no point of T is fixed by  $\pi_1(M)$ . Let S be a surface associated to the action of  $\pi_1(M)$  on T. Then there is a sequence

 $S = S_0, S_1, \dots, S_n$  of surfaces in M, each associated to the action, such that (i)  $S_{i+1}$  is obtained from  $S_i$  by a compression or by deleting the intersection of  $S_i$  with a trivial region for  $S_i$ ; and

(ii)  $S_n$  is essential.

*Proof.* A well-known finiteness result (cf. [Hem, *loc. cit.*]) asserts that, given any properly embedded surface S in a compact, irreducible 3-manifold, there exists a finite sequence of operations of the type described in (i) that replaces S by a (possibly empty) surface whose components are all incompressible and non-peripheral. If S is associated to the action of  $\pi_1(M)$  on T, Proposition 1.3.6 asserts that all the surfaces in the sequence are associated to the action. If the action has no fixed point then by 1.3.2 the final surface is non-empty; thus by definition it is essential.

PROPOSITION 1.3.8. Assume that no point of T is fixed by  $\pi_1(M)$ . Then there exists an essential surface in M associated to the action. Furthermore, if C is a connected subcomplex of  $\partial M$  such that the image of  $\pi_1(C)$  in  $\pi_1(M)$  is contained in a vertex stabilizer, then the surface may be taken to be disjoint from C.

Proof. Let  $\mathscr{K}$  be defined as in 1.3.3, and let  $\phi: M \to \mathscr{K}$  be a map inducing the natural isomorphism of fundamental groups (up to conjugacy). If C is a connected subcomplex of  $\partial M$  such that the image of  $\pi_1(C)$  in  $\pi_1(M)$  stabilizes a vertex s of the tree, then it follows from 1.3.5 that  $\phi$  may be taken to map Cinto  $\mathscr{K}_s$ . After a general-position homotopy we may assume that  $\phi$  is transverse to  $\mathscr{E}$ . Then by Proposition 1.3.4,  $S = \phi^{-1}(\mathscr{E})$  is a surface associated to the action. Clearly  $S \cap C = \emptyset$ . The conclusion now follows from Corollary 1.2.6 since the property of being disjoint from C is obviously preserved under compression and under deletion of components.

PROPOSITION 1.3.9. Let x be an ideal point of the curve  $\tilde{X}_0$  of Section 1.1. Let  $\alpha$  be a primitive element in L such that  $I_{e(\alpha)}(x) \neq \infty$ . Then either  $I_{e(\beta)}(x) \neq \infty$  for all  $\beta \in L$  or else  $\alpha$  is a strict boundary class.

*Proof.* Consider the action of  $\pi_1(M)$  on the tree associated to the ideal point x. By 1.2.6  $e(\alpha)$  stabilizes a vertex. Thus, by the proposition, there exists an essential surface S associated to the action which is disjoint from a simple closed curve in  $\partial M$  representing  $\alpha$ . It follows from Proposition 1.2.7 that S is not a fiber in any fibration of M over  $S^1$ . If  $\partial S$  is non-empty then  $\alpha$  is a strict boundary class. Otherwise, by 1.3.2,  $\pi_1(\partial M)$  stabilizes a vertex of T. Thus by 1.2.6 we have  $I_e(\beta) \neq \infty$  for all  $\beta$  in L.

#### 1.4. The norm

In this section we prove Proposition 1.1.2, which asserts that the degree of  $f_{\alpha}$  for  $\alpha \in L$  is given by a norm on the vector space  $V = H_1(\partial M; \mathbf{R})$ , and that the unit ball in this norm is a finite-sided polygon whose vertices are rational multiples of strict boundary classes. The proof is based on a study of the poles of the functions  $f_{\alpha}$ .

Let K denote the function field  $C(X_0) = C(\tilde{X}_0)$  of  $\tilde{X}_0$ . If f is an element of K and x is a point of  $\tilde{X}_0$  where f has a pole, then we denote the order of the pole by  $\prod_x(f)$ . If f does not have a pole at x we set  $\prod_x(f)$  equal to 0. Note that

degree 
$$f = \sum_{x \in \tilde{X}_0} \prod_x (f)$$
.

Note also that if  $\alpha \in L$  then the poles of  $f_{\alpha}$  all occur at ideal points of  $\tilde{X}_0$ .

LEMMA 1.4.1. For each ideal point x of  $\tilde{X}_0$  there is a homomorphism  $\phi_x: L \to \mathbb{Z}$  such that

$$\prod_{\mathbf{x}}(f_{\alpha}) = |\phi_{\mathbf{x}}(\alpha)|$$

for all  $\alpha \in L$ .

Proof. As in 1.2.4, we consider the tautological representation  $P: \pi_1(M) \to \operatorname{SL}_2(F)$ , where  $F = \mathbb{C}(R_0)$ . We regard F as an extension field of K. Note that  $P(\pi_1(\partial M))$  is an abelian subgroup of  $\operatorname{SL}_2(F)$ , and that for  $1 \neq \gamma \in \pi_1(\partial M)$ , trace  $P(\gamma) = I_{\gamma} \neq 2$  since  $I_{\gamma}$  is non-constant by 1.1.1. Hence there is a degree-two extension E of the field F such that P is equivalent (via an inner automorphism of  $\operatorname{GL}_2(E)$ ) to a representation  $P': \pi_1(M) \to \operatorname{SL}_2(E)$  which restricts to a diagonal representation of  $\pi_1(\partial M)$ . Let  $w: K^* \to \mathbb{Z}$  be the valuation defined by the ideal point x of  $\tilde{X}_0$ . By 1.2.2, there is a valuation  $v: E^* \to \mathbb{Z}$  such that  $v \mid K^* = d \cdot w$  for some positive integer d.

Let us fix a basis  $\alpha_1, \alpha_2$  for the lattice L. For i = 1 or 2, set

$$P(e(\alpha_i)) = \begin{bmatrix} \lambda_i & 0\\ 0 & \lambda_i^{-1} \end{bmatrix}$$

where the  $\lambda_i$  are elements of  $E^*$ . For any element  $\alpha = m\alpha_1 + n\alpha_2$  of L, we have

$$I_{e(\alpha)} = \text{trace } P(e(\alpha)) = \lambda_1^m \lambda_2^n + \lambda_1^{-m} \lambda_2^{-n}.$$

Hence

$$f_{\alpha} = I_{e(\alpha)}^2 - 4 = \left(\lambda_1^m \lambda_2^n - \lambda_1^{-m} \lambda_2^{-n}\right)^2.$$

But for any element g of E, g not equal to 0, 1 or -1, we have  $-\min(0, v(g - g^{-1})) = |v(g)|.$ 

Thus

$$\Pi_{x}(f_{\alpha}) = \Pi_{x}\left(\left(\lambda_{1}^{m}\lambda_{2}^{n} - \lambda_{1}^{-m}\lambda_{1}^{-n}\right)^{2}\right)$$
$$= -\min\left(0, w\left(\left(\lambda_{1}^{m}\lambda_{2}^{n} - \lambda_{1}^{-m}\lambda_{1}^{-n}\right)^{2}\right)\right)$$
$$= \frac{-2}{d}\min(0, v\left(\lambda_{1}^{m}\lambda_{2}^{n} - \lambda_{1}^{-m}\lambda_{1}^{-n}\right)\right)$$
$$= \frac{2}{d}|v\left(\lambda_{1}^{m}\lambda_{2}^{n}\right)|$$
$$= \frac{2}{d}|mv(\lambda_{1}) + nv(\lambda_{2})|.$$

The lemma follows upon setting  $\phi_x(\alpha) = (2/d)(mv(\lambda_1) + nv(\lambda_2))$ .

Now for each ideal point x of  $\tilde{X}_0$  we extend the homomorphism  $\phi_x$  given by 1.4.1 to a linear function  $\Phi_x$ :  $V \to \mathbf{R}$ . For  $\alpha \in V$  we set  $\|\alpha\| = \sum |\Phi_x(\alpha)|$ , where the sum is taken over all ideal points x of  $\tilde{X}_0$ . We shall show that  $\|\cdot\|$  is a norm on V and has the properties asserted in Proposition 1.1.2.

LEMMA 1.4.2. For each  $\alpha \in L$ ,  $\|\alpha\| = \text{degree } f_{\alpha}$ .

Proof. We have

degree 
$$f_{\alpha} = \sum_{x \in \tilde{X}_0} \prod_x (f_{\alpha}) = \sum_{x \text{ ideal}} \phi_x(\alpha) = ||\alpha||.$$

LEMMA 1.4.3.  $\|\cdot\|$  is a norm.

*Proof.* Since  $\|\cdot\|$  is a sum of absolute values of linear functions, it is sub-additive and homogeneous. Thus we need only show that  $\|\alpha\| \neq 0$  for  $\alpha \neq 0$ . Since the functions  $\Phi_x$  restrict to homomorphisms  $\phi_x \colon L \to \mathbb{Z}$ , it suffices to show that if  $\alpha$  is a non-zero element of L then  $\|\alpha\| \neq 0$ . Suppose that  $\alpha$  is a non-zero element of L then  $\|\alpha\| \neq 0$ . Suppose that  $\alpha$  is a non-zero element of L with  $\|\alpha\| = 0$ . Then by Lemma 1.4.2,  $f_{\alpha}$  would have degree zero, i.e. would be constant. Thus  $I_{e(\alpha)}$  would be constant, contradicting 1.1.1.

LEMMA 1.4.4. The unit ball for  $\|\cdot\|$  is a finite-sided polygon whose vertices are rational multiples of strict boundary classes in L.

*Proof.* Let J denote the set of all ideal points  $x \in \tilde{X}_0$  for which  $\phi_x$  is not identically zero. Then for  $\alpha \in V$  we have  $||\alpha|| = \sum_{x \in J} |\Phi_x(\alpha)|$ . Hence each vertex of the ball is contained in the zero set of one of the linear functions

 $\Phi_x, x \in J$ . But for  $x \in J$ ,  $\Phi_x$  vanishes exactly on a 1-dimensional subspace of V which is spanned by an element of L. Thus if  $\omega$  is a vertex of the unit ball then  $\omega = r\alpha$ , where  $\alpha$  is a primitive element of L such that  $\Phi_x(\alpha) = 0$  for some  $x \in J$ . For such an element  $\alpha$  we have  $\prod_x (f_\alpha) = 0$ , which implies that  $I_\alpha(x) \neq \infty$ . But by the definition of J there exists  $\beta \in L$  with  $\prod_x (f_\beta) = |\phi_x(\beta)| \neq 0$ , and hence  $I_\beta(x) = \infty$ . By 1.3.9 this implies that  $\alpha$  is a strict boundary class. Since  $||\alpha||$  is an integer, r must be rational.

Proposition 1.1.2 follows immediately from Lemmas 1.4.2, 1.4.3 and 1.4.4.

#### 1.5. Zeros at ordinary points

This section and the next are devoted to the proof of Proposition 1.1.3. Recall that we are given a primitive element  $\alpha$  of L which is not a strict boundary class. In addition we are given a point x of  $\tilde{X}_0$ , and a non-zero element  $\delta$  of L. We must show that if  $\pi_1(M(\alpha))$  is cyclic then  $Z_x(f_\alpha) \leq Z_x(f_\delta)$ . In this section we will treat the case where x is an ordinary point, and in Section 6 we will treat the case where x is an ideal point.

1.5.1. We denote by  $X_0^{\nu}$  the set of all ordinary points of  $\tilde{X}_0$ , and we let  $\nu$  denote the restriction to  $X_0^{\nu}$  of the birational equivalence  $\tilde{X}_0 \to X_0$ . Note that  $\nu$  is a regular map, i.e. is well-defined at every point of  $X_0^{\nu}$ . Thus we have a diagram:

$$\begin{array}{c}
R_{0} \\
\downarrow^{t} \\
K_{0}^{\nu} \xrightarrow{\nu} X_{0}
\end{array}$$

We shall prove:

PROPOSITION 1.5.2. Let x be a point of  $X_0^{\nu}$ , and assume that  $Z_x(f_{\alpha}) > Z_x(f_{\delta})$ . Then there exists a representation  $\rho \in R_0$  such that

(i)  $t(\rho) = \nu(x);$ 

(ii) the image in  $\mathrm{PSL}_2(\mathbb{C})$  of the group  $\rho(\pi_1(M)) \subset \mathrm{SL}_2(\mathbb{C})$  is non-cyclic; and

(iii)  $\rho(e(\alpha)) = \pm 1$ .

Note that if there exists a  $\rho$  satisfying (ii) and (iii) then it induces a homomorphism of  $\pi_1(M(\alpha)) = |\pi_1(M): \alpha = 1|$  onto a non-cyclic subgroup of PSL<sub>2</sub>(C), and thus  $\pi_1(M(\alpha))$  cannot be cyclic. Thus Proposition 1.5.2 implies Proposition 1.1.3 for the case of an ordinary point.

The proof of Proposition 1.5.2 will be based on a study of the normalization of the variety  $R_0$ .

1.5.3. Recall, for example from [Shf, Chapter II, Section 5], that a complex affine variety is said to be *normal* if its coordinate ring is integrally closed in its function field. If V is any complex affine variety, there is a complex affine variety  $V^{\nu}$  such that  $\mathbb{C}[V^{\nu}]$  is isomorphic to the integral closure of  $\mathbb{C}[V]$  in  $\mathbb{C}(V)$ . Thus  $V^{\nu}$  is normal. Furthermore, there is a natural regular birational map  $\nu: V^{\nu} \to V$  realizing the inclusion of  $\mathbb{C}[V]$  in its integral closure; and  $\nu$  is an isomorphism if and only if V is normal. The map  $\nu$  is finite-to-one [Shf, p. 116, Theorem 6]; hence for any subvariety Z of V, the irreducible components of  $\nu^{-1}(Z)$  each have the same dimension as Z.

These constructions are functorial in the sense that if  $\phi: V \to W$  is a dominating regular map of affine varieties then there is a unique regular map  $\phi^{\nu}: V^{\nu} \to W^{\nu}$  making the diagram



commute.

The singular set of a normal variety has codimension at least 2 [Shf, p. 111, Theorem 3]. In particular, a normal curve is smooth. Hence if V is an affine curve then  $V^{\nu}$  is the curve obtained from V by resolving its singularities; this justifies the notation of Proposition 1.5.2.

Using the normalization of  $R_0$  we can now complete the diagram of 1.5.1 to the commutative square below.



In order to prove Proposition 1.5.2 it clearly suffices to show that for every  $x \in X_0^{\nu}$  satisfying  $Z_x(f_{\alpha}) > Z_x(f_{\delta})$ , there is a point  $\tilde{\rho} \in R_0^{\nu}$  such that  $t^{\nu}(\tilde{\rho}) = x$  and such that  $\rho = \nu(\tilde{\rho})$  satisfies conditions (ii) and (iii) of 1.5.2. This assertion is implied by the next three results.

PROPOSITION 1.5.4. Let  $\alpha$  and  $\delta$  be non-zero elements of L. Suppose that x is a point of  $X_0^{\nu}$  such that  $Z_x(f_{\alpha}) > Z_x(f_{\delta})$ . Then for every  $\tilde{\rho} \in R_0^{\nu}$  with  $t^{\nu}(\tilde{\rho}) = x$ , the representation  $\rho = \nu(\tilde{\rho})$  satisfies  $\rho(e(\alpha)) = \pm 1$ .

**PROPOSITION 1.5.5.** For every point x of  $X_0^{\nu}$  there is a dense subset U of  $(t^{\nu})^{-1}(x)$  such that for every representation  $\rho \in \nu(U)$ , the image of  $\rho(\pi_1(M))$  in  $\text{PSL}_2(\mathbb{C})$  is noncyclic.

**PROPOSITION 1.5.6.** The map  $t^{\nu}$ :  $R_0^{\nu} \rightarrow X_0^{\nu}$  is surjective.

The remainder of this section is devoted to the proofs of Propositions 1.5.4, 1.5.5, and 1.5.6. The proof of Proposition 1.5.4 is based on the following algebraic lemma, which will also be used in Section 1.6.

LEMMA 1.5.7. Let v be a discrete valuation of a field F. Let A and B be commuting elements of  $SL_2^-(\mathcal{O}_v)$  such that

$$v((\operatorname{trace} A)^2 - 4) > v((\operatorname{trace} B)^2 - 4).$$

Then  $A \equiv \pm 1 \pmod{\mathcal{M}_v}$ .

**Proof.** Since A and B commute, there is an extension field E of degree at most 2 over F such that A and B have a common eigenvector  $\xi$  in  $E^2$ . By 1.2.3 we can extend v to a discrete valuation u of E. After multiplying  $\xi$  by an appropriate element of  $E^*$  we may assume that  $\xi$  is a primitive element of  $\mathcal{O}_u^2$ . Hence A and B are conjugate in  $\operatorname{GL}_2(\mathcal{O}_u)$  to matrices

 $\begin{bmatrix} a & x \\ 0 & a^{-1} \end{bmatrix}$  and  $\begin{bmatrix} b & y \\ 0 & b^{-1} \end{bmatrix}$ .

1.5.8. Since A and B commute we have  $ay + b^{-1}x = bx + a^{-1}y$ . Hence

$$(b - b^{-1})x = (a - a^{-1})y$$

On the other hand,  $(\operatorname{trace} A)^2 - 4 = (a - a^{-1})^2$  and  $(\operatorname{trace} B)^2 - 4 = (b - b^{-1})^2$ . Thus the hypothesis of the lemma implies that  $u(a - a^{-1}) > u(b - b^{-1})$ . With 5.8 this gives

$$u(x) \ge u(x) - u(y) = u(a - a^{-1}) - u(b - b^{-1}) > 0.$$

The inequality u(x) > 0 means that x is congruent to 0 modulo  $\mathcal{M}_u$ . We also have  $u(a - a^{-1}) > 0$ , which implies that  $a \equiv \pm 1 \pmod{\mathcal{M}_u}$ . It follows that A is congruent to  $\pm 1 \mod \mathcal{M}_u$  and therefore modulo  $\mathcal{M}_v$ .

Proof of Proposition 1.5.4. Consider a component Q of the algebraic set  $(t^{\nu})^{-1}(x) \subset R_0^{\nu}$  which passes through the point  $\tilde{\rho}$  of  $R_0^{\nu}$ . Since  $t^{\nu}$  is non-constant, Q has codimension 1 in  $R_0^{\nu}$ . Since  $R_0$  is normal, its singularities have codimension at least 2. In particular, a generic point of Q is a smooth point of  $R_0^{\nu}$ . Thus, by [Shf, p. 128], Q determines a discrete rank 1 valuation v of  $F = \mathbf{C}(R_0) = \mathbf{C}(R_0^{\nu})$ . A non-zero function in F is contained in  $\mathcal{O}_v$  if and only if

its divisor of poles does not contain Q; it is contained in  $\mathcal{M}_v$  if and only if its divisor of zeros does contain Q. Let  $K = \mathbb{C}(X_0) = \mathbb{C}(X_0^v)$  and let w denote the valuation of K determined by x. Since t(Q) = x, it is clear that  $\mathcal{O}_v \cap K = \mathcal{O}_w$ . Thus there is an integer d such that  $v|_{K^*} = d \cdot w$ .

1.5.9. Note that  $\mathbb{C}[R_0] \subset \mathbb{C}[R_0^{\nu}] \subset \mathcal{O}_{\nu}$ . Furthermore, for any  $h \in \mathbb{C}[R_0]$ , regarding h as a function on  $R_0$ , we have

$$h \in \mathcal{M}_{v} \Leftrightarrow h \circ v = 0 \text{ on } Q \Leftrightarrow h = 0 \text{ on } v(Q).$$

As in 1.2.3, we consider the tautological representation  $P: \pi_1(M) \to SL_2(F)$ . Recall that

$$f_{\alpha} = I_{e(\alpha)}^2 - 4 = (\text{trace } P(e(\alpha)))^2 - 4;$$

similarly  $f_{\delta} = (\text{trace } P(e(\delta)))^2 - 4$ . Since x is an ordinary point we have

$$Z_{x}(f_{\alpha}) = w(f_{\alpha}) = \frac{1}{d} \cdot v\big((\text{trace } P(e(\alpha)))^{2} - 4\big).$$

Likewise,

$$Z_{x}(f_{\delta}) = \frac{1}{d} \cdot v \big( (\text{trace } P(e(\delta)))^{2} - 4 \big).$$

Thus the hypothesis of Proposition 1.5.4 implies

$$v\big((\operatorname{trace} P(e(\alpha)))^2 - 4\big) > v\big((\operatorname{trace} P(e(\delta)))^2 - 4\big).$$

By Lemma 1.5.7 we have  $P(e(\alpha)) = \pm 1 \pmod{\mathcal{M}_v}$ .

By the definition of the tautological representation P, the entries of the matrix  $\rho(e(\alpha))$  are obtained by evaluating the entries of  $P(e(\alpha))$  at the point  $\rho \in R_0$ . The entries of  $P(e(\alpha))$  are elements of  $\mathbb{C}[R_0] \subset \mathcal{O}_v$ . Since  $P(e(\alpha)) \equiv \pm 1 \pmod{\mathcal{M}_v}$ , and since  $\rho \in \nu(Q)$ , observation 1.5.9 implies that  $\rho(e(\alpha)) = \pm 1$ .

The proof of Proposition 1.5.5 is based on:

LEMMA 1.5.10. Let x be a point of  $X_0$ . Let Z denote the set of all representations  $\rho \in t^{-1}(x)$  such that the image of  $\rho(\pi_1(M))$  in  $PSL_2(\mathbb{C})$  is cyclic. Then Z is contained in a countable union of algebraic subsets of  $t^{-1}(x)$ , each of dimension at most 2.

*Proof.* We consider the countable set  $\mathcal{N} = \{ \ker \rho: \rho \in Z \}$  of normal subgroups of  $\pi_1(M)$ . For each  $N \in \mathcal{N}$ , set  $Y_N = \{ \rho \in t^{-1}(x) | \rho(N) = \{1\} \}$ . Clearly each  $Y_N$  is a closed algebraic subset of  $t^{-1}(x)$ , and  $Z \subset \bigcup_{N \in \mathcal{N}} Y_N$ . We shall prove the lemma by showing that each  $Y_N$  has dimension at most 2.

Let  $N \in \mathcal{N}$  be given. Then  $\pi_1(M)/N$  has a central subgroup of order at most 2 with a cyclic quotient D. Let  $\gamma$  denote an element of  $\pi_1(M)$  which

projects to a generator of D. Then a representation  $\rho \in Y_N$  is uniquely determined by the matrix  $\rho(\gamma)$  and by the image of the central subgroup. The image of the central subgroup is either  $\{1\}$  or  $\{\pm 1\}$ . Furthermore, for any  $\rho \in Y_N$ , the trace of  $\rho(\gamma)$  is  $\tau_N = I_{\gamma}(x)$ . Thus there is a finite-to-one regular map from  $Y_N$  to the set of matrices of trace  $\tau_N$  in  $SL_2(\mathbb{C})$ . Since the latter set has dimension at most 2, so does  $Y_N$ .

Proof of Proposition 1.5.5. First note that  $R_0$  contains the discrete faithful representation  $\rho_0$ . Since int M has finite volume,  $\rho_0$  is irreducible. Hence by [C-S1, Corollary 1.5.3] we have dim  $R_0 - \dim X_0 = 3$ ; i.e. dim  $R_0 = 4$ . Since  $R_0^{\nu}$  is birationally equivalent to  $R_0$ , it also has dimension 4. It follows that for any point  $x \in X_0^{\nu}$ , the components of  $(t^{\nu})^{-1}(x)$  are each of dimension at least 3. (Actually, since  $R_0^{\nu}$  is irreducible and  $t^{\nu}$ :  $R_0^{\nu} \to X_0^{\nu}$  is non-constant, these components are all of dimension exactly 3.)

By Lemma 1.5.10, there is a countable collection of algebraic subsets  $A_1, A_2, \ldots$  of  $t^{-1}(\nu(x))$ , each of dimension at most 2, such that  $\bigcup_{i \ge 1} A_i$  contains every representation  $\rho$  for which the image of  $\rho(\pi_1(M))$  in PSL<sub>2</sub>(C) is cyclic. We set  $U = (t^{\nu})^{-1}(x) - \bigcup_{i \ge 1} \nu^{-1}(A_i)$ . By 1.5.10, each of the sets  $\nu^{-1}(A_i)$  has dimension at most 2. Hence U is dense in  $(t^{\nu})^{-1}(x)$ .

The proof of 1.5.6 is a refinement of the proof of Proposition 1.4.4 of [C-S1]. It is based on the following lemma, which is similar to a result deduced in [C-S1] from the "Burnside Lemma." Here we offer a more geometric proof which is extracted from [M-S1].

LEMMA 1.5.11. Let F be a field and let  $v: F^* \to \mathbb{Z}$  be a discrete valuation. Suppose that  $\Gamma$  is a finitely generated subgroup of  $\mathrm{SL}_2(F)$  such that for each  $g \in \Gamma$  trace  $g \in \mathcal{O}$ . Then  $\Gamma$  is conjugate in  $\mathrm{GL}_2(F)$  to a subgroup of  $\mathrm{SL}_2(\mathcal{O})$ .

**Proof.** (cf. [M-S1, II.3.17]). By considering the rational canonical form, one sees that each element of  $\Gamma$  is conjugate in  $\operatorname{GL}_2(F)$  to an element of  $\operatorname{SL}_2(\mathcal{O})$ . Thus, by 2.1, under the action of  $\operatorname{SL}_2(F)$  on the tree T determined by the valuation v, each element of  $\Gamma$  fixes a vertex. According to [Se, Cor. 3 to Prop. 26], if a finitely generated group acts on a tree so that each element fixes a vertex, then there is a vertex which is fixed by the entire group. Thus  $\Gamma$  stabilizes a vertex of the tree T, and hence by 1.2.1 is conjugate in  $\operatorname{GL}_2(F)$  to a subgroup of  $\operatorname{SL}_2(\mathcal{O})$ .

We will also need an elementary fact from algebraic geometry. A valuation v of the function field  $\mathbf{C}(V)$  of an affine variety V will be said to be *supported in* V if  $\mathbf{C}[V] \subset \mathcal{O}_v$ .

1.5.12. Recall that for any valuation v of a field F,  $\mathcal{O}_{v}$  is integrally closed in F. Hence a valuation of  $\mathbf{C}(V) = \mathbf{C}(V^{\nu})$  is supported in V if and only if it is supported in  $V^{\nu}$ .

**LEMMA** 1.5.13. Suppose that  $\phi: V \to C$  is a non-constant regular map from a complex affine variety to a complex affine curve. Regard  $\mathbf{C}(V)$  as an extension of C(C). Let x be a smooth point of C and let w be the valuation determined by x. Assume that w extends (in the sense of Section 2) to a discrete valuation of  $\mathbf{C}(V)$  supported in V. Then x is in the image of  $\phi$ .

*Proof.* Note that  $\mathscr{I} = \mathscr{M}_{w} \cap \mathbb{C}[C]$  is the ideal of functions in  $\mathbb{C}[C]$  that vanish at x. Hence the algebraic subset  $\phi^{-1}(x)$  of V is defined by the ideal  $\mathscr{I} \cdot \mathbb{C}[V]$ . Hence in order to prove that x is in the image of  $\phi$ , it is enough to show that  $\mathscr{I} \cdot \mathbb{C}[V]$  is a proper ideal in  $\mathbb{C}[V]$ .

By hypothesis, w extends to a valuation v of C(V) which is supported in V. We have  $\mathbf{C}[V] \subset \mathcal{O}_{v}$  and  $\mathscr{I} \subset \mathscr{M}_{v} \subset \mathscr{M}_{v}$ . Hence  $\mathscr{I} \cdot \mathbf{C}[V]$  is contained in  $\mathscr{M}_{v}$ and is therefore proper.  $\square$ 

Proof of Proposition 1.5.6. Let x be a point of  $X_0^{\nu}$  and let w denote the valuation of  $C(X_0) = C(X_0^{\nu})$  determined by x. By Theorem 1.2.3, w extends to some valuation v of  $C(R_0)$ . As in 1.2.4, we consider the tautological representation  $P: \pi_1(M) \to SL_2(F)$ . For each  $\gamma \in \pi_1(M)$  we have

trace 
$$P(\gamma) = I_{\gamma} \in \mathbb{C}[X_0] \subset \mathcal{O}_w \subset \mathcal{O}_v.$$

Hence by Lemma 1.5.11, there is an element A of  $GL_2(F)$  such that  $A \cdot P(\pi_1(M)) \cdot A^{-1} \subset SL_2(\mathcal{O}_{\mu})$ . Using A we define a rational map  $\psi: R_0 \to R_0$ as follows. If  $\alpha$  is an element of  $\operatorname{GL}_2(\mathbb{C}_{\nu})$ . Using if we define a rational map  $\psi$ .  $R_0 \to R_0$ as follows. If  $\alpha$  is an element of  $\operatorname{GL}_2(\mathbb{C})$  then let  $i_{\alpha}$  denote the automorphism of  $\operatorname{SL}_2(\mathbb{C})$  given by  $i_{\alpha}(\xi) = \alpha \xi \alpha^{-1}$ . If  $A = \begin{bmatrix} a & b \\ c & d \end{bmatrix}$  then for  $\rho \in R_0$  let  $A(\rho)$ denote the matrix  $\begin{bmatrix} a(\rho) & b(\rho) \\ c(\rho) & d(\rho) \end{bmatrix} \in \operatorname{GL}_2(\mathbb{C})$ . We then set  $\psi(\rho) = i_{A(\rho)} \circ \rho$ . Since  $\nu$ :  $R_0^{\nu} \to R_0$  is birational,  $\psi$  determines a rational map  $\psi^{\nu}$ :  $R_0^{\nu} \to R_0^{\nu}$ .

Moreover, we have the following commutative diagrams of rational maps:



Let L denote the subvariety of  $R_0^{\nu}$  which is the closure of the image under  $\psi^{\nu}$  of the set of regular points of  $\psi^{\nu}$ . Thus  $\psi^{\nu}$  co-restricts to a dominating rational map  $\Psi: R_0^{\nu} \to L$ . We will show that  $x \in t^{\nu}(L)$ ; this will complete the proof that  $t^{\nu}$  is surjective.

This content downloaded from 128.248.155.225 on Sun, 4 Jan 2015 21:37:06 PM All use subject to JSTOR Terms and Conditions

Since  $\Psi$  is dominating, it induces a homomorphism  $\mathbf{C}(L) \to \mathbf{C}(R_0)$  which may be used to pull the valuation v back to a valuation u on  $\mathbf{C}(L)$ . The commutative diagram



gives rise to the commutative diagram



of function fields. This shows that u is an extension of w. Thus by Lemma 1.5.13 we need only show that u is supported in L.

Note that the coordinate ring  $\mathbb{C}[R_0]$  is generated by the entries of the matrices  $P(\gamma)$  for  $\gamma \in \pi_1(M)$ . Since  $i_A(P(\pi_1(M))) \subset \mathrm{SL}_2(\mathcal{O}_v)$ , it follows that  $\psi^*(\mathbb{C}[R_0]) \subset \mathcal{O}_v$ . Since a discrete valuation ring is integrally closed we have  $\psi^*(\mathbb{C}[R_0^v]) \subset \mathcal{O}_v$ . In particular  $\Psi^*(\mathbb{C}[L]) \subset \mathcal{O}_v$ , or equivalently  $\mathbb{C}[L] \subset \mathcal{O}_u$ .  $\Box$ 

#### 1.6. Zeroes at ideal points

In this section we complete the proof of Proposition 1.1.3. In Section 1.5 we gave the proof in the case where the point  $x \in \tilde{X}_0$  in the hypothesis of 1.1.3 is an ordinary point. In the case where x is an ideal point of  $\tilde{X}_0$ , the conclusion of 1.1.3 follows immediately from the following result:

PROPOSITION 1.6.1. Let x be an ideal point of  $\tilde{X}_0$ . Let  $\alpha$  and  $\delta$  be non-zero elements of L. Suppose that  $\alpha$  is primitive and is not a strict boundary class, and that

 $Z_{\mathbf{x}}(f_{\alpha}) > Z_{\mathbf{x}}(f_{\delta}).$ 

Then there is a closed surface in M which is incompressible in  $M(\alpha)$ . In particular  $\pi_1(M(\alpha))$  is not cyclic.

The proof of Proposition 1.6.1 will occupy this entire section. Throughout the section, we consider a fixed ideal point x and fixed elements  $\alpha$  and  $\delta$  of L which satisfy the hypotheses of 1.6.1.

1.6.2. In particular we have  $Z_x(f_{\alpha}) > 0$ , and hence  $I_{e(\alpha)}(x)$  is finite. Since  $\alpha$  is not a strict boundary class, it follows from Proposition 1.3.9 that  $I_{e(\beta)}(x) \neq \infty$  for all  $\beta \in L$ .

Next we consider an action of  $\pi_1(M)$  on a tree T associated to the ideal point x. Recall that by 1.2.6, no point of T is fixed by  $\pi_1(M)$ . Thus, it follows

from 1.3.2 that any surface associated to the action is non-empty. Since for every  $g \in \pi_1(\partial M)$  we have  $I_g(x) \neq \infty$ , each element of  $\pi_1(\partial M)$  fixes some point of T. Hence by [Se, Cor. 3 to Prop. 26], there is one vertex of T which is fixed by  $\pi_1(\partial M)$ . It therefore follows from Proposition 1.3.8 that there is a closed surface associated to the action of  $\pi_1(M)$  on T.

The method of proof of 1.6.1 is to show that in the set of all closed surfaces in M which are associated to the action of  $\pi_1(M)$  on T, there is one which is incompressible in  $M(\alpha)$ .

If S is a closed surface then we define

$$\chi_{-}(S) = \sum - \chi(S_i)$$

where the sum is taken over all components  $S_i$  of S which are not 2-spheres. We denote by #(S) the number of components of S. The *complexity* of S is the pair  $c(S) = (\chi_{-}(S), \#(S))$ , which will be regarded as an element of the set  $N \times N$  of pairs of non-negative integers with the lexicographic ordering.

1.6.3. Among all closed surfaces associated to the action of  $\pi_1(M)$  on T we fix one, S, of minimal complexity. The rest of this section will be devoted to proving that S is incompressible in  $M(\alpha)$ . This will complete the proof of Proposition 1.6.1.

The first step is easy:

LEMMA 1.6.4. The surface S is essential in M. In particular every component of S has genus greater than 1.

*Proof.* We have observed that a surface associated to the action of  $\pi_1(M)$  on T is non-empty. If some component of S is a 2-sphere or a boundary-parallel surface, then it follows from 1.3.6 that there is a surface which is associated to the action and which is obtained from S by deleting one or more components. This contradicts the minimality of c(S).

Now suppose that there is a compressing disk D for S. By 1.3.6, the surface S' obtained by compressing S along D is associated to the action. If the component  $S_0$  of S containing  $\partial D$  has genus greater than 1, then  $\chi_-(S') < \chi_-(S)$ , and again the minimality of c(S) is contradicted. If  $S_0$  is a torus then c(S') = c(S), and some component of S' is a 2-sphere. By 1.3.6 we can then produce an associated surface by deleting one or more components of S. Again this contradicts the minimality of c(S).

This proves that S is essential. Since M contains no essential tori, it follows that every component of S has genus greater than 1.

The main step in the proof that S is incompressible in  $M(\alpha)$  uses the following strong property of the action of  $\pi_1(M)$  on T.

LEMMA 1.6.5. Let s be any vertex of T which is fixed by  $\pi_1(\partial M) \subset \pi_1(M)$ . Then the element  $e(\alpha)$  of  $\pi_1(\partial M)$  acts trivially on the link of s in T.

(Note that if the subgroup  $\pi_1(\partial M)$  and the element  $\alpha$  are conjugated by an element of  $\pi_1(M)$ , the truth of the above statement is unaffected. Thus the statement is consistent with the conventions of 1.0.2.)

*Proof.* Let w denote the valuation of  $K = \mathbb{C}(X_0)$  defined by the ideal point x of  $\tilde{X}_0$ . Since by 1.6.2,  $I_{\alpha}(x)$  and  $I_{\delta}(x)$  are finite, we have  $w(f_{\alpha}) = Z_x(f_{\alpha})$  and  $w(f_{\delta}) = Z_x(f_{\delta})$ . According to 1.2.5, the tree T is defined in terms of some extension v of w to  $F = \mathbb{C}(R_0)$ . There is a positive integer d such that  $v|K^* = d \cdot w$ .

The action of  $\pi_1(M)$  on T is defined by the tautological representation  $P: \pi_1(M) \to \operatorname{SL}_2(F)$ . The vertices of T are equivalence classes of  $\mathcal{O}_v$ -lattices in  $F^2$ . Since  $\operatorname{GL}_2(F)$  acts transitively on the vertices of T, we may assume after conjugation by an element of  $\operatorname{GL}_2(F)$  that the given vertex s is the class of the standard lattice  $\mathcal{O}_v^2$ . By 1.2.1 the stabilizer of this vertex in  $\operatorname{SL}_2(F)$  is  $\operatorname{SL}_2(\mathcal{O}_v)$ . Thus the hypothesis of the lemma implies that  $P(\pi_1(\partial M)) \subset \operatorname{SL}_2(\mathcal{O}_v)$ .

The hypothesis of Proposition 1.6.1 gives

$$v((\operatorname{trace} P(e(\alpha)))^2 - 4) = d \cdot w(I_{\alpha}^2 - 4)$$
  
=  $d \cdot w(f_{\alpha})$   
=  $d \cdot Z_x(f_{\alpha})$   
>  $d \cdot Z_x(f_{\delta}) = v((\operatorname{trace} P(e(\delta)))^2 - 4).$ 

Since  $P(e(\alpha))$  and  $P(e(\delta))$  commute, Lemma 1.5.7 implies  $P(e(\alpha)) \equiv \pm 1 \pmod{\mathcal{M}_v}$ . By 1.2.2 this is equivalent to saying that  $e(\alpha)$  acts trivially on the link of s.

1.6.6. Proof of Proposition 1.6.1. As observed earlier it suffices to show that S is incompressible in  $M(\alpha)$ . The proof will proceed by contradiction. Assume that S is compressible in  $M(\alpha)$ . Since S has no sphere components by Lemma 1.6.4, there must exist a compressing disk D for S in  $M(\alpha)$ .

Recall that  $M(\alpha)$  is obtained from M by attaching a solid torus J along  $\partial M$  so that the boundary of a meridian disk in J represents the homology class  $\alpha$  in  $\partial M$ . We may suppose D to be chosen so that each component of  $D \cap J$  is a meridian disk. Thus  $P = D \cap M$  is a planar surface whose boundary components are  $\omega = \partial D$  and simple closed curves  $\alpha_1, \ldots, \alpha_k$  representing the homology class  $\alpha$  in  $\partial M$ .

We shall denote by N the closure of the component of  $M(\alpha) - S$  that contains  $\partial M$ . Let V be a regular neighborhood of D in  $M(\alpha)$  which meets J in

a regular neighborhood of  $D \cap J$  in J. Consider the surface Q in  $M(\alpha)$  obtained by compressing S along D, using the neighborhood V: Explicitly we have  $Q = (S \cup \partial V) - int(S \cap \partial V).$ 

1.6.7. Since by Lemma 1.6.4, every component of S has negative Euler characteristic, and since Q is obtained from S by a compression, we have  $\chi_{-}(Q) < \chi_{-}(S)$ . Set  $Q^{-} = Q \cap M$ . Thus  $Q^{-}$  is obtained from S by replacing an annular neighborhood of  $\omega$  by two parallel copies of P.

Claim 1.6.8. The surface  $Q^- \subset M$  is associated to the action of  $\pi_1(M)$  on T.

Proof of claim. We consider the complex  $\mathscr{K}$  determined by the action as in 1.3.3. Recall that there is a natural isomorphism between the groups  $\pi_1(M)$ and  $\pi_1(\mathscr{K})$  of covering transformations. By Proposition 1.3.4 there is a map  $\phi: M \to \mathscr{K}$  which is transverse to  $\mathscr{E}$  with  $\phi^{-1}(\mathscr{E}) = S$ , and which induces the natural isomorphism up to conjugation.

We now fix base points  $q \in \omega \subset S \subset N \subset M$  and  $r = \phi(q) \in \mathscr{K}$ . Any choice of lifts  $\tilde{q}$  and  $\tilde{r}$  of q and r determines identifications of  $\pi_1(M)$  with  $\pi_1(M, q)$  and of  $\pi_1(\mathscr{K})$  with  $\pi_1(\mathscr{K}, r)$ . We may choose  $\tilde{q}$  and  $\tilde{r}$  so that in terms of these identifications,  $\phi_*: \pi_1(M, q) \to \pi_1(\mathscr{K}, r)$  is the natural isomorphism between  $\pi_1(M)$  and  $\pi_1(\mathscr{K})$ . By 1.3.5, there is an edge e of T with  $\phi(S) \subset \mathscr{K}_e$ , and  $\pi_1(\mathscr{K}_e, r) = \pi_e$ . There is a vertex s incident to e such that  $\phi(N) \subset \mathscr{K}_s$ ; again by 1.3.5 we have  $\pi_1(\mathscr{K}_s, r) = \pi_s$ .

Also  $\partial M \subset N$ . Hence up to conjugacy in  $\pi_1(N, q)$ , there is a natural identification of  $\pi_1(\partial M)$  with a subgroup of  $\pi_1(N, q)$ . Thus  $\phi_*(\pi_1(\partial M))$  is a subgroup of  $\pi_1(\mathscr{K}_s, r)$  defined up to conjugacy in  $\pi_1(\mathscr{K}_s, r)$ . Since  $\pi_1(\mathscr{K}_s, r)$  fixes the vertex s of T, it follows from Lemma 1.6.5 that every conjugate of  $\phi_*(e(\alpha))$  in  $\pi_1(\mathscr{K}_s, r)$  acts trivially on the link of s.

All but one of the boundary components of the planar surface P are curves in  $\partial M \subset N$  which represent the conjugacy class of  $e(\alpha)$  in  $\pi_1(N, q)$ . Hence if  $\theta: P \to N$  denotes the restriction of  $\phi$ , then  $\theta_*(\pi_1(P, q))$  is contained in the normal closure of  $\phi_*(e(\alpha))$  in  $\pi_1(\mathscr{K}_s)$ . Therefore  $\theta_*(\pi_1(P, q))$  acts trivially on the link of s. In particular,  $\theta_*(\pi_1(P, q)) \subset \pi_e = \pi_1(\mathscr{K}_e, r)$ . Since  $\mathscr{K}_s$  is aspherical, this implies that there exists a homotopy  $\Theta: P \times [0, 1] \to \mathscr{K}_s$ , constant on  $\omega$ , such that  $\Theta_0 = \theta$  and  $\Theta_1(P) \subset \mathscr{K}_e$ . Such a homotopy  $\Theta$  can clearly be extended to a homotopy  $\Phi: M \times [0, 1] \to \mathscr{K}$ , constant outside a small neighborhood of Vin N, such that  $\Phi_0 = \phi$  and  $\Phi_1^{-1}(\mathscr{K}_e) = S \cup \partial V$ . Furthermore,  $\Phi_1$  can be perturbed to a map  $\psi: M \to \mathscr{K}$ , transverse to  $\mathscr{E}$ , such that  $\psi^{-1}(\mathscr{E}) = Q$ . Since  $\psi$  is homotopic to  $\phi$  it then follows from Proposition 1.3.4 that Q is associated to the action.  $\Box$  We now apply Corollary 1.3.7 to the surface  $Q^-$ . This gives a sequence of surfaces  $Q^- = Q_0^-$ ,  $Q_1^-$ , ...,  $Q_n^-$ , each associated to the action of  $\pi_1(M)$  on T. Each  $Q_{i+1}^-$  is obtained from  $Q_i^-$  by compressing along a disk or deleting certain components. Furthermore,  $Q_n^-$  is essential in M. For each i we clearly have  $\partial Q_i^- \subset \partial Q^-$ .

In particular, any boundary component of  $Q_n^-$  must represent the homology class  $\alpha$  in L. Note also that, since  $Q_n^-$  is associated to the action of  $\pi_1(M)$  on T, it follows from Proposition 1.2.7 that  $Q_n^-$  cannot be a fiber in a fibration of M over S<sup>1</sup>. Since  $\alpha$  is not a strict boundary class it follows that  $Q_n^-$  is closed.

Next we define a closed surface  $Q_i$  in  $M(\alpha)$  by adding meridian disks of J to the boundary components of  $Q_i^-$ . We have  $Q_0 = Q$  and  $Q_n = Q_n^-$ .

Note also that  $Q_{i+1}$  is obtained from  $Q_i$  either by surgery on a (possibly trivial) simple closed curve or by deleting certain components. Hence

$$\chi_{-}(Q_n) \leq \chi_{-}(Q_{n-1}) \leq \cdots \chi_{-}(Q_0) = \chi_{-}(Q)$$

which with 1.6.7 gives  $\chi_{-}(Q_n) < \chi_{-}(S)$ , and hence  $c(Q_n) < c(S)$ . But  $Q_n = Q_n^-$  is a closed surface associated to the action. This contradicts the minimality of c(S). Hence the assumption in 1.6.6 that S is compressible in  $M(\alpha)$  is false, and the proof of Proposition 1.6.1 is complete.

#### Chapter 2

#### 2.0. Introduction

Let M be a compact, connected, irreducible 3-manifold such that  $\partial M$  is a torus. In light of Theorem 1.0.1, we shall be concerned in this chapter with the case of the Cyclic Surgery Theorem in which either M contains an essential torus or one of r, s is a strict boundary slope. To state the theorem that we shall use to deal with the first possibility, we need the following standard definition. Let V be a solid torus. A *cable space* is the complement of an open tubular neighborhood of a (p, q)-cable of the core of V, where p, q are coprime integers with  $q \ge 2$ .

THEOREM 2.0.1. If M contains an essential torus S which compresses in M(r) and M(s), then either  $\Delta(r, s) \leq 1$  or S and  $\partial M$  cobound a cable space in M.

A closed, connected 3-manifold is a *Haken manifold* if it is irreducible and contains an incompressible surface. A *lens space* is a closed 3-manifold of Heegaard genus 1 whose fundamental group is finite (cyclic). (Thus  $S^3$  and  $S^1 \times S^2$  are excluded.)

The next two theorems, together with Addendum 2.0.4, establish the Cyclic Surgery Theorem in the case where one of the given slopes is a strict boundary slope.

THEOREM 2.0.2. Suppose that dim  $H_1(M; \mathbf{Q}) > 1$ . If M(r) and M(s) have cyclic first homology groups and are not Haken manifolds then  $\Delta(r, s) \leq 1$ .

THEOREM 2.0.3. Suppose that dim  $H_1(M; \mathbf{Q}) = 1$ . If r is a boundary slope, then either

(i) M(r) is a Haken manifold; or

(ii) M(r) is a connected sum of two lens spaces; or

(iii) M contains a closed incompressible surface which remains incompressible in M(s) whenever  $\Delta(r, s) > 1$ ; or

(iv) M fibers over  $S^1$  with fiber a planar surface having boundary slope r.

We remark that possibility (iv) was overlooked in the statement of Proposition 2 of the announcement [CGLS]. Note that in that case M(r) is homeomorphic to  $S^1 \times S^2$ , which has cyclic fundamental group. However, we have the following addendum to Theorem 2.0.3.

Addendum 2.0.4. In case (iv) of Theorem 2.0.3, if r is a strict boundary slope then conclusion (iii) holds also.

The chapter is organized as follows. In Sections 2.1, 2.2 and 2.3 we take the first steps towards proving Theorem 2.0.3. We start with an essential surface F in M having boundary slope r, and consider the corresponding closed surface  $\hat{F}$  in M(r). We show that if F is suitably chosen, then either  $\hat{F}$  is an incompressible surface of positive genus and conclusion (i) holds; or  $\hat{F}$  is a separating 2-sphere which decomposes M as a connected sum of two lens spaces, giving conclusion (ii); or  $\hat{F}$  is a non-separating 2-sphere and conclusion (iv) holds; or M contains a closed essential surface S, disjoint from F, with certain additional properties. The main tool used is a mild extension of a result of Jaco [J] giving conditions under which the addition of a 2-handle to a 3-manifold yields a manifold with incompressible boundary. This work is begun in Section 2.1, and completed in Section 2.2, in the case that F has positive genus, and in Section 2.3, in the case that F is planar.

To complete the proof of Theorem 2.0.3, we need to show that the closed surface S remains incompressible in M(s) whenever  $\Delta(r, s) > 1$ . More generally, we consider the question: For which slopes r does a given closed essential surface S in M compress in M(r)? To study this, we assume that there exist compressing disks for S in M(r) and M(s), say, and carry out a graph-theoretic analysis of the intersection of the two corresponding planar surfaces in M. In the case that S is a torus, this leads to Theorem 2.0.1. (An earlier result along these lines was obtained by Litherland [L], who used the same general approach.) A very similar argument proves Theorem 2.0.2. Here, if M(r) and M(s) are not Haken manifolds, then, since they have positive first Betti number, they each contain a 2-sphere which does not bound a 3-ball, and we apply the same analysis to the intersection of the two planar surfaces in M corresponding to these 2-spheres. Again, an earlier result along these lines was proved in [Gr-L]. By refining the graph-theoretic analysis used to prove Theorems 2.0.1 and 2.0.2, we obtain information on the general question of when a closed essential surface in M compresses in M(r). In particular, making use of the special properties of the surface S that arises in the proof of Theorem 2.0.3, we show that it satisfies conclusion (iii).

Section 2.4 contains the statements of the results that we prove on the question of when a closed essential surface compresses under Dehn surgery. We also show in that section how Theorems 2.0.1 and 2.0.2 follow from these results. In Section 2.5 we set up the graph-theoretic machinery that we will use in the proofs, and establish some basic connections between the graph theory and the topology. We also state the main graph-theoretic propositions that we require, and show how they imply the theorems stated in Section 2.4 and enable us to complete the proof of Theorem 2.0.3, as well as prove Addendum 2.0.4. The main graph-theoretic analysis is carried out in Section 2.6, in which we prove the propositions stated in Section 2.5. After Theorems 1.0.1, 2.0.1, 2.0.2 and 2.0.3, and Addendum 2.0.4, the proof of the Cyclic Surgery Theorem reduces to a consideration of manifolds which contain cable spaces. This is discussed in Section 2.7. Finally, in Section 2.8, we give a proof of Corollary 7.

#### 2.1. Dehn surgery along a boundary slope

In this and the next two sections M will be a compact, connected, irreducible 3-manifold, such that  $\partial M$  is a torus and dim  $H_1(M; \mathbf{Q}) = 1$ .

We wish to analyze M(r) in the case that r is a boundary slope. To do this we consider an essential surface F in M with boundary slope r which is minimal in an appropriate sense, and study the handle decompositions of certain associated submanifolds of M and M(r). We begin this analysis in the present section, and refine it in Sections 2.2 and 2.3, which treat the cases that F is non-planar and planar respectively. Specifically, we will show that either F is non-planar and conclusion (i) of Theorem 2.0.3 holds, or F is planar and either conclusion (ii) or (iv) of Theorem 2.0.3 holds, or M contains a closed incompressible surface S with certain properties. Sections 2.4, 2.5, and 2.6 are devoted largely to showing that this surface S remains incompressible in M(s) whenever  $\Delta(r, s) > 1$ . This is conclusion (iii) of Theorem 2.0.3. It will be convenient to introduce the following notation. If S is a surface, and  $C_1, \ldots, C_n$  are disjoint simple loops in S, then  $\sigma(S; \bigcup_{i=1}^n C_i)$  will denote the surface resulting from surgery on S along  $C_1, \ldots, C_n$ .

If Q is a 3-manifold, and  $C_1, \ldots, C_n$  are disjoint simple loops in  $\partial Q$ , then  $\tau(Q; \bigcup_{i=1}^n C_i)$  will denote the 3-manifold obtained by attaching 2-handles to Q along disjoint regular neighborhoods of  $C_1, \ldots, C_n$ .

Note that if  $C_1, \ldots, C_n \subset S \subset \partial Q$ , then  $\sigma(S; \bigcup_{i=1}^n C_i) \subset \partial \tau(Q; \bigcup_{i=1}^n C_i)$ .

The following handle addition lemma will play a key role in our analysis of M(r).

LEMMA 2.1.1. Let Q be an irreducible 3-manifold, S a surface in  $\partial Q$  which is compressible in Q, and C a simple loop in S such that S – C is incompressible in Q. Suppose that  $\sigma(S; C)$  is not a 2-sphere. Then  $\tau(Q; C)$  is irreducible and  $\sigma(S; C)$  is incompressible in  $\tau(Q; C)$ .

In the form stated, this is proved in [C-G]. See also [J], [Jo], [P], [Sch1], and [D-S].

By a compression body we shall mean a cobordism W (rel  $\partial$ ) between surfaces  $\partial_+ W$  and  $\partial_- W$  such that  $W \cong \partial_+ W \times I \cup 2$ -handles  $\cup$  3-handles and  $\partial_- W$  has no 2-sphere components. It follows that W is irreducible and  $\partial_- W$  is incompressible in W. If Q is an irreducible 3-manifold and F is a surface in  $\partial Q$ , then there is a maximal compression body  $W \subset Q$  with  $\partial_+ W = F$ , which is unique up to isotopy. (See [Bo, §2].) The *inner boundary* of W is  $F^- =$  $\partial_- W \cup \partial F \times I$ . Thus  $\partial F^- = \partial F$ , and  $F^-$  is incompressible in Q.

There is a disjoint union of disks  $(D^*, \partial D^*) \subset (Q, F)$  such that, for a regular neighborhood  $N(D^*)$  of  $D^*$  compatible with a collar  $\partial Q \times I$  of  $\partial Q$  in Q, we have  $W = F \times I \cup N(D^*) \cup 3$ -handles. Note that  $F^- \cong \sigma(F; \partial D^*)$  with all 2-sphere components removed.

Our assumption that dim  $H_1(M; \mathbf{Q}) = 1$  implies that  $H_2(M) = 0$ . It follows that if  $(G, \partial G) \subset (M, \partial M)$  is a non-separating surface then  $[\partial G] \neq 0$  in  $H_1(\partial M)$ , and hence, by successively tubing adjacent oppositely oriented components of  $\partial G$  and then compressing, we obtain an essential non-separating surface  $(G, \partial G) \subset (M, \partial M)$  such that  $\partial G \neq \emptyset$  and all components of  $\partial G$ , when given the orientations induced by some orientation of G, are homologous on  $\partial M$ .

Now let r be a boundary slope on  $\partial M$ , and let  $(F, \partial F) \subset (M, \partial M)$  be an essential separating surface with boundary slope r, such that each component of F has non-empty boundary, and such that the number of components of  $\partial F$  is minimal subject to these conditions. Thus F either is connected or has exactly two components, each of which is non-separating. In the latter case we shall assume (as we may without loss of generality) that F consists of two parallel copies of a non-separating surface G, which (by the argument given in the

previous paragraph) necessarily has all its boundary components oriented coherently on  $\partial M$ . Note that  $[F, \partial F] = 0$  in  $H_2(M, \partial M)$ .

We shall refer to the two cases described above as the *connected* and *disconnected* cases, respectively. They will for the most part be treated in parallel; the main divergence occurs when F is planar.

Note that the disconnected case can only occur when r is the unique slope that corresponds to an element in the kernel of the map  $H_1(\partial M; \mathbf{Q}) \to H_1(M; \mathbf{Q})$ .

The manifold obtained by cutting M along F has two components, X and X', say. (In the disconnected case, X' (say) is homeomorphic to  $G \times I$ .) The number of components of  $\partial F$  is even, say  $2n \ (n \ge 1)$ , and these components cut  $\partial M$  into parallel annuli  $A_1, \ldots, A_n, A'_1, \ldots, A'_n$  such that  $\partial X = F \cup \bigcup_{i=1}^n A_i$ ,  $\partial X' = F \cup \bigcup_{i=1}^n A'_i$ . Note that each of  $\partial X$  and  $\partial X'$  has, in the connected case, genus f + n, where f is the genus of F, and in the disconnected case, genus 2g + (n - 1), where g is the genus of G. Let  $C_i, C'_i$  be a core of  $A_i, A'_i$  respectively. Regarding the solid torus J as the union of 2n 2-handles with attaching regions  $A_1, \ldots, A_n, A'_1, \ldots, A'_n$ , we see that  $M(r) = M \cup J \cong \tau(X; \bigcup_{i=1}^n C_i) \bigcup_{\hat{F}} \tau(X'; \bigcup_{i=1}^n C_i') = \hat{X} \bigcup_{\hat{F}} \hat{X}'$ , say, where  $\hat{F} = \partial \hat{X} = \partial \hat{X}'$  is the closed surface obtained from F by capping off  $\partial F$  with meridian disks of J.

For  $1 \le i \le n$ , let  $F_i = F \cup A_i \subset \partial X$ ; so  $\partial F_i = \bigcup_{j \ne i} \partial A_j$ . Let  $V_i$  be a maximal compression body for  $F_i$  in X. The inner boundary of  $V_i$ ,  $F_i^-$ , is incompressible in X. Since  $M = X \bigcup_F X'$  and F is incompressible in M, it follows that  $F_i^-$  is also incompressible in M. Since the number of boundary components of  $F_i^-$  is 2(n-1), our assumption on F implies that every separating component of  $F_i^-$  with non-empty boundary is a boundary-parallel annulus. Furthermore, since  $[F_i^-, \partial F_i^-] = [F_i, \partial F_i] = [F, \partial F] = 0$  in  $H_2(M, \partial M)$  (and  $H_2(M, \partial M) \cong \mathbb{Z}$ ), the number of non-separating components of  $F_i^-$  is even. It follows that if  $F_i^-$  has a non-separating component then it has one with at most (n-1) boundary components. Since, as noted above, any non-separating surface in M has non-empty boundary, taking two parallel copies of such a component would contradict our minimality assumption on F. We conclude that  $F_i^-$  consists of (n-1) annuli, each parallel into  $\partial M$ , together with some closed surface (possibly empty) in int X. Let B be one of the annuli. Since B separates M, and int  $B \cap F_i = \emptyset$ , and  $F_i$  is connected, it follows that  $\partial B = \partial A_i$  for some  $j \neq i$ . We may therefore number the annuli  $B_i^{(j)}$ ,  $j \neq i$ , so that  $\partial B_i^{(j)} = \partial A_i \ (j \neq i).$ 

LEMMA 2.1.2. For  $1 \le i \le n$ , there exist disjoint disks  $E_i^{(j)}$  in  $X, j \ne i$ , such that  $\partial E_i^{(j)}$  meets  $C_j$  transversely in a single point and is disjoint from  $C_k$  if  $k \ne i$  or j.

*Proof.* Fix  $i, 1 \le i \le n$ , and recall the annuli  $B_i^{(j)}, j \ne i$ . Each  $B_i^{(j)}$  is parallel into  $\partial M$ . Let  $U_i$  be the solid torus realizing this parallelism. Then  $U_i \subset X$ 

(otherwise  $F \subset U_j$ ),  $\partial U_j = A_j \cup B_i^{(j)}$ , and the  $U_j$  are disjoint. Choose a meridian disk of  $U_j$  with boundary  $\alpha_j \cup \beta_j$ , where  $\alpha_j$ ,  $\beta_j$  are spanning arcs of the annuli  $A_j$ ,  $B_i^{(j)}$  respectively. Now  $V_i \cong F_i \times I \cup 2$ -handles  $\cup$  3-handles; dually,  $V_i \cong \partial_- V_i \times I \cup 0$ -handles  $\cup$  1-handles. An isotopy of  $\beta_j$  (rel  $\partial$ ) in  $B_i^{(j)}$  will move it off the disks that constitute the attaching regions of the 1-handles, and then a further isotopy (rel  $\partial$ ) in  $V_i$  (using the product structure of  $\partial_- V_i \times I$ ) will take it to an arc  $\beta'_j$  in  $F_i \subset \partial X$ . A corresponding isotopy and extension of the chosen meridian disk of  $U_j$  gives a disk  $E_i^{(j)}$  in X with  $\partial E_i^{(j)} = \alpha_j \cup \beta'_j$ . These disks  $E_i^{(j)}$ ,  $j \neq i$ , satisfy the conditions stated.

Let  $W_i$  be the (possibly punctured) compression body in X with  $\partial_+ W_i = \partial X$ defined by  $W_i = \partial X \times I \cup N(\bigcup_{j \neq i} E_i^{(j)})$ , where  $N(\bigcup_{j \neq i} E_i^{(j)})$  is a suitable regular neighborhood of  $\bigcup_{j \neq i} E_i^{(j)}$ . Thus  $\partial W_i$  is the disjoint union of  $\partial X$  and  $\partial_- W_i$ , where  $\partial_- W_i$  is a (closed) connected surface which, in the connected case, has genus f + 1, and, in the disconnected case, genus 2g. Hence  $W_i$  is a compression body unless F is planar and disconnected, in which case  $X = W_i \cup 3$ -ball is a handlebody of genus n - 1.

LEMMA 2.1.3. For  $1 \leq i \leq n$ ,  $\tau(W_i, \bigcup_{i \neq i} C_i) \cong \partial_- W_i \times I$ .

*Proof.* This follows by canceling the 2-handle corresponding to  $C_j$  with  $E_i^{(j)}, j \neq i$ .

For  $0 \le k \le n$ , let  $X_k = \tau(X; \bigcup_{i=1}^k C_i)$ . Thus  $X_0 = X$ , and  $X_n = \hat{X}$ . We shall apply the following lemma with k = n - 1 in Section 2.2, and with k = n - 2 in Section 2.3.

LEMMA 2.1.4. If F is either non-planar or connected then, for  $0 \le k \le n-1$ ,  $X_k$  is irreducible and  $\partial X_k - \bigcup_{i=k+1}^n C_i$  is incompressible in  $X_k$ .

*Proof.* We prove this by induction on k, using Lemma 2.1.1. The assertion holds for k = 0, since X is irreducible and F is incompressible in X. So suppose  $1 \le k \le n - 1$ , and that the assertion holds for k - 1. Thus  $X_{k-1}$  is irreducible and  $\partial X_{k-1} - \bigcup_{i=k}^{n} C_i = (\partial X_{k-1} - \bigcup_{i=k+1}^{n} C_i) - C_k$  is incompressible in  $X_{k-1}$ . Write  $S_k = \partial X_{k-1} - \bigcup_{i=k+1}^{n} C_i$ . We shall apply Lemma 2.1.1 with  $Q = X_{k-1}$ ,  $S = S_k$ , and  $C = C_k$ . Since  $X_k = \tau(X_{k-1}; C_k)$ , this will establish the conclusion of the lemma for k, provided we show that  $S_k$  is compressible in  $X_{k-1}$ .

To do this, let  $(D_k^*, \partial D_k^*) \subset (X, F_k)$  be a disjoint union of disks such that the maximal compression body  $V_k$  for  $F_k$  in X can be expressed as  $F_k \times I \cup N(D_k^*)$  for a suitable regular neighborhood  $N(D_k^*)$  of  $D_k^*$ . Since  $\partial D_k^* \cap C_i = \emptyset$ ,  $i \neq k$ , we have  $(D_k^*, \partial D_k^*) \subset (X_{k-1}, S_k)$ . We claim that some component of  $\partial D_k^*$  is essential in  $\partial X_{k-1}$ , and a fortiori in  $S_k$ . For if not, then  $\sigma(\partial X_{k-1}; \partial D_k^*)$  would be homeomorphic to the disjoint union of  $\partial X_{k-1}$  with some 2-spheres. However,  $\partial X_{k-1}$  is a connected surface which has, in the connected case, genus  $f + n - (k - 1) \ge f + 2 \ge 2$ , and, in the disconnected case, genus  $2g + (n - 1) - (k - 1) \ge 2g + 1 \ge 3$  (since  $g \ge 1$  here by hypothesis). On the other hand,  $\sigma(\partial X; \partial D_k^*)$  contains n - 1 tori  $T_k^{(j)}$ , corresponding to the tori  $A_j \cup B_k^{(j)}$ ,  $j \ne k$ , and therefore  $\sigma(\partial X_{k-1}; \partial D_k^*) = \sigma(\sigma(\partial X; \partial D_k^*); \bigcup_{j=1}^{k-1} C_j)$  contains  $\bigcup_{j=k+1}^n T_k^{(j)}$ , a disjoint union of tori which is non-empty since  $k \le n - 1$ . This contradiction completes the proof of the lemma.

We note for completeness that if F is disconnected and planar then  $X_{n-1}$  is a 3-ball.

#### 2.2. The non-planar case

As a first approximation to the conclusion of Theorem 2.0.3 in the case that F is non-planar, we have the following proposition.

PROPOSITION 2.2.1. Suppose that F is non-planar. Then either

(i) M(r) is irreducible and  $\hat{F}$  is incompressible in M(r); or

(ii) M contains a closed incompressible surface S which is disjoint from F.

We shall ultimately show that the surface S remains incompressible in M(s) whenever  $\Delta(r, s) > 1$ . Our proof of this will require some additional properties of S. These are contained in the following addendum to Proposition 2.2.1.

Addendum 2.2.2. In conclusion (ii) of Proposition 2.2.1, we may assume that the surface S cobounds with  $\partial M$  a manifold N in M such that either

(i) N contains an annulus with one boundary component in S and the other having slope r in  $\partial M$ ; or

(ii) (a) N(r) is irreducible; and

(b) there is a compressing disk for S in N(r) which misses  $\hat{F}$ ; and

(c) no compressing disk as in (b) meets only a single component of  $\partial M - \partial F$ .

Proof of Proposition 2.2.1. Recall the notation established in Section 2.1, and suppose that  $\partial_{-} W_{n}$  is compressible in X. Since  $\tau(W_{n}; \bigcup_{i=1}^{n-1}C_{i}) \cong \partial_{-} W_{n} \times I$ by Lemma 2.1.3, it follows that  $\partial X_{n-1}$  is compressible in  $X_{n-1}$ . By Lemma 2.1.4 (with k = n - 1),  $X_{n-1}$  is irreducible and  $\partial X_{n-1} - C_{n}$  is incompressible in  $X_{n-1}$ . Hence, by Lemma 2.1.1,  $\hat{X} = \tau(X_{n-1}; C_{n})$  is irreducible and  $\hat{F}$  is incompressible in  $\hat{X}$ . Similarly, if  $\partial_{-} W'_{n}$  is compressible in X', then  $\hat{X}'$  is irreducible and  $\hat{F}$  is incompressible in  $\hat{X}'$ . Therefore  $M(r) = \hat{X} \cup_{\hat{F}} \hat{X}'$  is irreducible and  $\hat{F}$ is incompressible in M(r), which is conclusion (i).

Otherwise,  $\partial_{-} W_n$  (say) is incompressible in X, and hence in  $M = X \bigcup_F X'$ , giving conclusion (ii).

270

The argument above in the case that n = 1 and F is disconnected is the same as that given in [J]. Also, the case in which n = 1, F is connected and X, X' are handlebodies is treated in [P].

Let  $\hat{W}_n = \tau(W_n; \bigcup_{i=1}^n C_i)$ . The observations contained in the following lemma will be needed in the proof of Addendum 2.2.2.

LEMMA 2.2.4. If F is non-planar then

- (a)  $\hat{W}_n$  is irreducible;
- (b)  $\hat{F}$  is incompressible in  $\hat{W}_n$ ; and
- (c)  $\partial_{-} W_{n}$  is compressible in  $\hat{W}_{n}$ .

*Proof.* By Lemma 2.1.3,  $\tau(W_n; \bigcup_{i=1}^{n-1}C_i) \cong \partial_- W_n \times I$ , and hence  $\hat{W}_n \cong \tau(\partial_- W_n \times I; C_n)$ . Also,  $C_n$  is essential in  $\partial X_{n-1} = \partial_- W_n \times \{0\}$ , since each component of  $\partial X$  cut along  $\bigcup_{i=1}^n C_i$  (which is homeomorphic to F) is non-planar. Hence  $\hat{W}_n$  is a compression body with a single 2-handle whose attaching circle is essential. Conclusions (a), (b) and (c) follow.

We remark that if F is planar and connected, then  $\partial_- W_n$  is a torus and  $C_n$  is essential in  $\partial_- W_n \times \{0\}$ , since  $C_1, \ldots, C_n$  are homologically independent in  $\partial X$ . Hence  $\hat{W}_n$  is a punctured solid torus, so that conclusion (c) still holds. We shall use this in Section 2.3.

Proof of Addendum 2.2.2. Recall that conclusion (ii) of Proposition 2.2.1 arises when (without loss of generality)  $\partial_{-} W_{n}$  is incompressible in X. We distinguish two subcases.

(2.A) ∂<sub>-</sub> W<sub>n</sub>' is incompressible in X';
(2.B) ∂<sub>-</sub> W<sub>n</sub>' is compressible in X'.

(In the disconnected case, (2.B) always holds, since X' is a handlebody.)

First note that since  $\partial_- W_n$  is incompressible in X,  $W_n$  is the unique (up to isotopy) maximal compression body for  $\partial X$  in X. Since  $\partial_- W_i$  is homeomorphic to  $\partial_- W_n$  for all  $i, 1 \le i \le n$ , it follows that all the  $W_i$  are isotopic. Let us write  $W = W_i$ ,  $S_- = \partial_- W_i$ . Similarly, in case (2.A) above, let  $W' = W'_i$ ,  $S'_- = \partial_- W'_i$ .

In case (2.A), let  $S = S_{-} \cup S'_{-}$ , and  $N = W \bigcup_{F} W'$ . In case (2.B), let  $S = S_{-}$  and  $N = W \bigcup_{F} X'$ .

In case (2.A),  $N(r) = \hat{W} \bigcup_{\hat{F}} \hat{W}'$ , which is irreducible by Lemma 2.2.3. In case (2.B),  $N(r) = \hat{W} \bigcup_{\hat{F}} \hat{X}'$ , which is irreducible by Lemma 2.2.3 and the fact that  $\hat{X}'$  is irreducible and  $\hat{F}$  is incompressible in  $\hat{X}$  (see the proof of Proposition 2.2.1). Thus (ii)(a) of the addendum is satisfied.

By Lemma 2.2.4,  $S_{-}$  is compressible in  $\hat{W}$ . This shows, in both cases (2.A) and (2.B) that (ii)(b) holds.

Finally, consider (ii)(c). First suppose that we are in case (2.B). Any compressing disk for  $S = S_{-}$  in N(r) which misses  $\hat{F}$  lies in  $\hat{W}$ . If there were such a disk which met only a single component of  $\partial M - \partial F$ , then there would be a planar surface in W with one boundary component essential in  $S_{-}$  and all other boundary components in a single annulus  $A_k$ , say. Hence  $S_{-}$  would compress in  $\tau(W; C_k) = \tau(W_i; C_k), 1 \le i \le n$ . If n > 1, we could choose  $i \ne k$ , and thereby obtain a contradiction to Lemma 2.1.3. Therefore n = 1. But in that case, W is just a collar  $\partial X \times I$  of  $\partial X$  in X, and condition (i) of the addendum is satisfied by taking the annulus  $C_1 \times I \subset \partial X \times I \subset N$ . In case (2.A), we additionally apply the same argument to  $S'_{-}$  in W', showing that again either  $S = S_{-} \cup S'_{-}$  satisfies condition (ii)(c) or n = 1 and S satisfies condition (i).

Remark. It is actually easier in case (2.A) to show that S remains incompressible in M(s) if  $\Delta(r, s) > 1$ , than in case (2.B). Specifically, this follows from Theorem 2.4.5, which in turn is a direct consequence of the graph-theoretic Proposition 2.5.6, and the latter is easier to prove than the corresponding graph-theoretic proposition (Proposition 2.5.9) that is needed to handle case (2.B). However, since the argument in case (2.B) also applies to case (2.A), for ease of organization we have chosen not to separate the two cases.

#### 2.3. The planar case

The purpose of this section is to prove the following proposition. Recall that if F is disconnected it consists of two copies of a non-separating surface G.

PROPOSITION 2.3.1. Suppose that F is planar. In the connected case, either

(i)  $\hat{F}$  decomposes M(r) as a connected sum of two lens spaces; or

(ii) M contains an incompressible torus which is disjoint from F and which compresses in M(r).

In the disconnected case,

(iii) M fibers over  $S^1$  with fiber G.

For the connected case we shall need the following lemma.

LEMMA 2.3.2. Let  $C_1, C_2$  be disjoint simple loops in the boundary of a handlebody X of genus 2, with the property that there exist disks  $E_1, E_2$  in X such that  $C_i$  intersects  $\partial E_i$  transversely in a single point, i = 1, 2. Then either

(i)  $\tau(X; C_1 \cup C_2)$  is a punctured lens space; or

(ii)  $\partial X - C_1 \cup C_2$  is compressible in X.

To prove Lemma 2.3.2, we first prove a purely algebraic lemma, which is an easy consequence of Nielsen's "sign condition" for a primitive element of a free group of rank 2 [N].

[] will denote the conjugacy class of an element in a group. A set of elements  $\{x_1, \ldots, x_n\}$  of a free group F is a *basis-up-to-conjugacy* for F if there exists a basis  $\{a_1, \ldots, a_n\}$  for F such that  $[x_i] = [a_i], 1 \le i \le n$ . Recall that an element of a free group F is *primitive* if it belongs to some basis for F. Finally,  $\langle \rangle$  denotes normal closure.

LEMMA 2.3.3. Let x and y be primitive elements of a free group F of rank 2. Then either

- (i)  $H_1(F/\langle x, y \rangle)$  is cyclic of order  $k, 1 < k < \infty$ ; or
- (ii)  $\{x, y\}$  is a basis-up-to-conjugacy; or
- (iii)  $[x] = [y^{\pm 1}].$

*Proof.* Since x is primitive, F has a basis of the form  $\{x, z\}$ . Let w be a cyclically reduced word in x, z such that [y] = [w]. By [N], since y is primitive, all the exponents of z in w have the same sign. Hence either conclusion (i) holds or z occurs at most once in w, with exponent  $\pm 1$ . Therefore, we may take w to be either  $x^n$  or  $x^n z^{\pm 1}$ , for some integer n. In the first case we must have  $n = \pm 1$  (since y is primitive), and we obtain conclusion (ii). In the second case, since  $\{x, x^n z^{\pm 1}\}$  is a basis for F, we obtain conclusion (ii).

Proof of Lemma 2.3.2. First note that since  $C_1$  and  $\partial E_1$  (say) are dual,  $\tau(X;C_1 \cup C_2)$  is obtained by attaching a 2-handle to a solid torus.

The loops  $C_1$  and  $C_2$  (when oriented) determine conjugacy classes  $[C_1]$ ,  $[C_2]$  in  $\pi_1(X)$ . The existence of the disk  $E_i$  implies that  $[C_i]$  is the conjugacy class of a primitive element of  $\pi_1(X)$ , i = 1, 2. Therefore Lemma 2.3.3 applies.

If case (i) of Lemma 2.3.3 holds, then we obtain conclusion (i) of Lemma 2.3.2.

In cases (ii) and (iii) of Lemma 2.3.3, we use the result of Zieschang [Z] that the minimal geometric length of a system of disjoint simple loops in the boundary of a handlebody is equal to its minimal algebraic length. (The geometric (resp. algebraic) length is computed from the cyclic words (resp. cyclic reduced words) that record the intersections of the loops with the boundaries of a complete system of meridian disks for the handlebody. For further discussion and a short proof of Zieschang's theorem, see [Wa2]. For an even shorter proof, see [K].) In the present situation, this implies that there exist disjoint disks  $D_1$ ,  $D_2$  in X, such that X cut along  $D_1 \cup D_2$  is a 3-ball, with the property that, in case (ii) of Lemma 2.3.3,  $C_i$  meets  $\partial D_i$  transversely in a single point and is disjoint from  $\partial D_j$ ,  $i = 1, 2, j \neq i$ , and in case (iii), each of  $C_1$  and  $C_2$  meets  $\partial D_1$  transversely in a single point and is disjoint form  $\partial X - C_1 \cup C_2$  is compressible in X, giving conclusion (ii).  $\Box$ 

Proof of Proposition 2.3.1. (a) The connected case. Recall the surfaces  $\partial_{-} W_{n} \subset X$ ,  $\partial_{-} W_{n}' \subset X'$  defined in Section 2.1. Since F is planar these are tori.

First suppose that  $\partial_{-} W_{n}$  (say) is incompressible in X. Then M contains an incompressible torus which is disjoint from F. Also,  $\partial_{-} W_{n}$  compresses in  $\hat{W}_{n} = \tau(W_{n}; \bigcup_{i=1}^{n} C_{i})$  (see the remark after the proof of Lemma 2.2.3) and hence in M(r). We thus obtain conclusion (ii).

So we suppose that  $\partial_- W_n$  compresses in X (and that  $\partial_- W'_n$  compresses in X'). Since X is irreducible,  $\partial_- W_n$  bounds a solid torus in X, implying that X is a handlebody of genus n. It is convenient to distinguish two cases:

(1) n = 1. Here F is an annulus and X is a solid torus. Since F is essential in  $M = X \bigcup_F X'$ , the image of  $H_1(F)$  in  $H_1(X)$ , which coincides with the image of  $H_1(C_1)$  in  $H_1(X)$ , has index k, say, where  $1 < k < \infty$ . Therefore  $\hat{X} = \tau(X; C_1)$ is a punctured lens space whose fundamental group has order k.

(2) n > 1. Recall Lemma 2.1.2. By canceling the 2-handle corresponding to  $C_j$  with the disk  $E_n^{(j)}$ ,  $j = 1, \ldots, n-2$ , we see that  $X_{n-2}$  is a handlebody of genus 2. Consider the disjoint simple loops  $C_{n-1}, C_n \subset \partial X_{n-2}$ . Since the boundaries of the disks  $E_n^{(n-1)}$  and  $E_{n-1}^{(n)}$  are disjoint from  $C_1, \ldots, C_{n-2}$ , we have  $E_n^{(n-1)}, E_{n-1}^{(n)} \subset X_{n-2}$ . Also,  $\partial E_n^{(n-1)}$  (resp.  $\partial E_{n-1}^{(n)}$ ) intersects  $C_{n-1}$  (resp.  $C_n$ ) transversely in a single point. Application of Lemma 2.1.4 with k = n - 2 shows that  $\partial X_{n-2} - C_{n-1} \cup C_n$  is incompressible in  $X_{n-2}$ . Lemma 2.3.2 now implies that  $\hat{X}$  is a punctured lens space.

Similarly (in both cases (1) and (2)), since  $\partial_- W'_n$  is compressible in  $X', \hat{X}'$  is also a punctured lens space, and we obtain conclusion (i) of the proposition. (b) The disconnected case.

(1) n = 1. Here G is a disk and M is a solid torus, so that we have conclusion (iii).

(2) n > 1. Recall from Section 2.1 that here X is a handlebody of genus n - 1. Also, the disks  $\{E_i^{(j)}: j \neq i\}$ , for  $1 \leq i \leq n$ , given by Lemma 2.1.2, show that the set of loops  $\{C_j: j \neq i\}$  is primitive (in the sense of [Gr2]) for each i,  $1 \leq i \leq n$ . Therefore, by [Gr2, Proposition 2.1],  $(X, G) \cong (G \times I, G \times \{0\})$ . Recalling that  $(X', G) \cong (G \times I, G \times \{0\})$  also, we see that M fibers over  $S^1$  with fiber G.

We remark that as far as the Cyclic Surgery Theorem is concerned, the cases n = 1 of both (a) and (b) are actually excluded by the assumption that M is not Seifert fibered. However, as there is no need to exclude them in the statement of Proposition 2.3.1 or Theorem 2.0.3, we do not do so.

#### 2.4. Reduction and boundary reduction under Dehn surgery

Throughout this section M will be an irreducible 3-manifold (not necessarily compact) with a torus boundary component T. If r is a slope on T, M(r) will denote the manifold obtained by attaching a solid torus to M along T so that the boundary of a meridian disk of the solid torus has slope r. Let S be a surface in

 $\partial M - T$  which is incompressible in M. We are interested in the question of when S compresses in M(r). Note that the case in which S is a closed incompressible surface in int M can be reduced to the situation just described by cutting M along S.

We make the following conjecture.

Conjecture 2.4.1. If S compresses in  $M(r_1)$  and  $M(r_2)$  then  $\Delta(r_1, r_2) \leq 1$ , unless M contains an incompressible annulus with one boundary component in S and the other in T.

Examples showing the need to rule out annuli are provided by the cable spaces (see the remarks immediately following the statement of Theorem 2.4.4 below). Examples (with S a torus) where S compresses in  $M(r_i)$  for two slopes  $r_1, r_2$  (with  $\Delta(r_1, r_2) = 1$ ) have been given by Berge (unpublished) and Gabai [Ga]. In fact an example is given in [Ga] where S compresses in  $M(r_i)$  for three slopes  $r_1, r_2, r_3$  (with  $\Delta(r_i, r_i) = 1$ ,  $i \neq j$ ).

In the direction of Conjecture 2.4.1, we are able to prove the following.

THEOREM 2.4.2. If S compresses in  $M(r_1)$  and  $M(r_2)$ , then either

(i)  $\Delta(r_1, r_2) \le 2; \text{ or }$ 

(ii) *M* contains an annulus with one boundary component in S and the other having slope  $r_0$  in *T*, where  $\Delta(r_0, r_i) = 1$ , i = 1, 2; or

(iii) M is homeomorphic to  $T \times I$ .

To complete the proof of Theorem 2.0.3, after Propositions 2.2.1 and 2.3.1, it would suffice to prove Conjecture 2.4.1. This can be achieved when S is a torus (see Theorem 2.4.4 below). In particular, this completes the proof of Theorem 2.0.3 when F is planar. In the non-planar case, the completion of the proof of Theorem 2.0.3 may be described as a modification of the proof of Theorem 2.4.2 which makes use of the special properties of the surface S that are described in Addendum 2.2.2. (Thus Theorem 2.4.2 itself is not used in the proof of the Cyclic Surgery Theorem, but we include it for its interest as a general result on the problem of when surfaces compress under Dehn surgery.)

If conclusion (ii) of Theorem 2.4.2 holds, then it is possible to obtain more explicit information about the compressibility of S in M(r). This is described in the following theorem.

THEOREM 2.4.3. Suppose that M contains an annulus with one boundary component in S and the other having slope  $r_0$  in T. Then

(a) S compresses in  $M(r_0)$ .

(b) S is incompressible in M(r) if  $\Delta(r, r_0) > 1$ , unless M is homeomorphic to  $T \times I$ .

(c) If  $\Delta(r_i, r_0) = 1$ , i = 1, 2, then  $M(r_1) \cong M(r_2)$ . In particular, S compresses in M(r) for some r such that  $\Delta(r, r_0) = 1$  if and only if it compresses in M(r) for all such r.

To state the improvement that can be made to Theorem 2.4.2 when S is a torus, recall the definition of a *cable space* given in Section 2.0.

THEOREM 2.4.4. If S is a torus which compresses in  $M(r_1)$  and  $M(r_2)$ , then either

- (i)  $\Delta(r_1, r_2) \le 1$ ; or
- (ii) M is a cable space; or
- (iii) M is homeomorphic to  $T \times I$ .

The examples mentioned above shows that this result is best possible.

If M is a cable space, with boundary components S and T (since there is an automorphism of M interchanging S and T, the labeling is irrelevant), then in fact S compresses in M(r) for infinitely many slopes r. This is accounted for by the existence of an incompressible annulus in M connecting the two boundary components, as described in Theorem 2.4.3(c) (see [Gr-L, Remark on p. 125]). The cable spaces thus provide simple examples to illustrate Theorem 2.4.3 and to show the need to exclude annuli in the formulation of Conjecture 2.4.1.

Another situation in which Theorem 2.4.2 can be improved is the following.

THEOREM 2.4.5. If  $S_1$  and  $S_2$  are disjoint surfaces in  $\partial M - T$ , each incompressible in M, such that  $S_i$  compresses in  $M(r_i)$ , i = 1, 2, then  $\Delta(r_1, r_2) \leq 1$ .

The above theorems can be regarded as restricting the boundary slopes in T of essential punctured disks in M. The next result deals with the analogous question for punctured spheres.

THEOREM 2.4.6. If  $(P_i, \partial P_i) \subset (M, T)$  is an essential planar surface with boundary slope  $r_i$ , i = 1, 2, then either

- (i)  $\Delta(r_1, r_2) \le 1$ ; or
- (ii)  $M(r_1)$  or  $M(r_2)$  contains a lens space as a connected summand.

An immediate corollary is the following.

COROLLARY 2.4.7. If  $M(r_1)$  and  $M(r_2)$  are reducible, then either

- (i)  $\Delta(r_1, r_2) \le 1$ ; or
- (ii)  $M(r_1)$  or  $M(r_2)$  contains a lens space as a connected summand.

Corollary 2.4.7 will be used to prove Theorem 2.0.2.

Theorem 2.4.6 and Corollary 2.4.7 may be compared with [Gr-L, Theorem 1.1], where it is shown that the alternative conclusions (i) and (ii) can be replaced by the single conclusion that  $\Delta(r_1, r_2) \leq 4$ . Probably none of these results is best possible.

In Section 2.5 we shall show how the theorems stated above follow from certain graph-theoretic propositions; these in turn will be proved in Section 2.6.

The remainder of this section is devoted to showing how Theorems 2.0.1 and 2.0.2 follow quickly from Theorems 2.4.4 and 2.4.5, and Corollary 2.4.7, respectively.

**Proof of Theorem 2.0.1.** Let S be an essential torus in int M which compresses in M(r) and M(s). Let M' be the component of the manifold obtained by cutting M along S that contains  $\partial M$ . There are two cases.

(1) S does not separate M. Then  $\partial M' = S_1 \cup S_2 \cup \partial M$ , where  $S_1$  and  $S_2$  are copies of S. If some  $S_i$  compresses in both M'(r) and M'(s), then Theorem 2.4.4 implies that  $\Delta(r, s) \leq 1$ . On the other hand, if  $S_1$  (say) compresses in M'(r) while  $S_2$  compresses in M'(s), then Theorem 2.4.5 implies that  $\Delta(r, s) \leq 1$ .

(2) S separates M. Then  $\partial M' = S \cup \partial M$ . If S compresses in M'(r) and M'(s), then by Theorem 2.4.4 either  $\Delta(r, s) \leq 1$  or M' is a cable space. (Conclusion (iii) of Theorem 2.4.4 is impossible since S is essential in M.)

Proof of Theorem 2.0.2. Since dim  $H_1(M; \mathbf{Q}) > 1$ , we have that dim  $H_1(M(r); \mathbf{Q}) \ge 1$  for all slopes r. Hence M(r) is a Haken manifold unless it is reducible. By Corollary 2.4.7, M(r) and M(s) reducible implies that either  $\Delta(r, s) \le 1$  or M(r) (say) contains a lens space as a connected summand. But the latter possibility is ruled out by the assumption that  $H_1(M(r))$  is cyclic.  $\Box$ 

# 2.5. The basic graph-theoretic analysis of reduction and boundary reduction

In this section we describe the basic machinery with which the theorems stated in Section 2.4 will be proved. In particular, we show how these theorems, and the remaining part of Theorem 2.0.3, follow from certain graph-theoretic propositions.

Throughout, we shall use the indices  $\alpha$  and  $\beta$  to denote 1 or 2, with the convention that, when they are used together,  $\{\alpha, \beta\} = \{1, 2\}$ .

Let M and S be as described at the beginning of Section 2.4, and let  $D_{\alpha}$  be a compressing disk for S in  $M(r_{\alpha}) = M \cup J_{\alpha}$ . We may assume that  $D_{\alpha}$  meets the solid torus  $J_{\alpha}$  in a disjoint union of meridian disks of  $J_{\alpha}$ . (Note that  $D_{\alpha} \cap J_{\alpha}$  is necessarily non-empty since S is incompressible in M.) Then  $P_{\alpha} = D_{\alpha} \cap M$  is a planar surface, with one outer boundary component  $\partial_0 P_\alpha = \partial D_\alpha$ , lying in S, and  $n_\alpha$ , say, inner boundary components  $\partial_x P_\alpha$ ,  $x = 1, ..., n_\alpha$ , each having slope  $r_\alpha$  in T. We shall always assume that  $P_\alpha$  is incompressible in M. This is guaranteed if  $D_\alpha$  is chosen so that  $n_\alpha$  is minimal (for example, over all compressing disks for S in  $M(r_\alpha)$ ), and we shall usually make this choice. On the one occasion when this may not be possible (this occurs for  $P_1$  in the final step of the proof of Theorem 2.0.3), it will be easy to see that we may still assume that  $P_1$  is incompressible in M.

By an isotopy of  $P_2$ , say, we may assume that  $P_1$  and  $P_2$  meet in general position. Note that  $\partial_0 P_\alpha$  meets no inner boundary component of  $P_\beta$ . Then  $P_1 \cap P_2 = \mathscr{A} \perp \mathscr{B} \perp \mathscr{C}$ , where  $\mathscr{A}$  is a disjoint union of arcs with at least one endpoint in an inner boundary component of  $P_\alpha$ ,  $\mathscr{B}$  is a disjoint union of arcs with both endpoints in  $\partial_0 P_\alpha$ , and  $\mathscr{C}$  is a disjoint union of simple closed curves. By a standard innermost disk argument, a further isotopy of  $P_2$  will ensure that no component of  $\mathscr{C}$  bounds a disk in  $P_\alpha$ , since  $P_\beta$  is incompressible and M is irreducible. Finally, again by an isotopy of  $P_2$ , say, we may assume that each inner boundary component of  $P_\alpha$  meets each inner boundary of  $P_\beta$  in  $\Delta = \Delta(r_1, r_2)$  points. In fact, if the inner boundary components of  $P_\beta$  so that they are consecutive on T, then going round each inner boundary component of  $P_\alpha$  (in some direction) the indices of the inner boundary components of  $P_\beta$  that we encounter are, in order,  $1, \ldots, n_\beta, \ldots, 1, \ldots, n_\beta$ (repeated  $\Delta$  times).

A graph in a disk D will consist of a finite number of vertices in int D, together with a finite number of edges. Each edge either connects a vertex to a (possibly non-distinct) vertex, or connects a vertex to  $\partial D$ . The former are *interior edges*, the latter, *boundary edges*. We assume that the endpoints on  $\partial D$  of all boundary edges are distinct.

Let  $\Gamma_{\alpha}$  be the graph in the disk  $D_{\alpha}$  obtained by taking the union of  $\mathscr{A}$  with the cone (in the corresponding meridian disk of  $J_{\alpha}$ ) on  $\mathscr{A} \cap \partial_x P_{\alpha}$  for each inner boundary component  $\partial_x P_{\alpha}$  of  $P_{\alpha}$ . Thus the vertices x of  $\Gamma_{\alpha}$  are in one-to-one correspondence with the inner boundary components  $\partial_x P_{\alpha}$  of  $P_{\alpha}$ , and the edges of  $\Gamma_{\alpha}$  are in one-to-one correspondence with the components of  $\mathscr{A}$ . Note that each vertex has valency  $\Delta n_{\beta}$ . Let e be an edge of  $\Gamma_{\alpha}$ , with one of its endpoints the vertex x, say. Then e corresponds to an arc in  $\mathscr{A}$  with one endpoint on  $\partial_x P_{\alpha}$  which is a point of intersection of  $\partial_x P_{\alpha}$  with some inner boundary component  $\partial_y P_{\beta}$ , say, of  $P_{\beta}$ . We say that e has label y at x. In this way, each incidence of an edge of  $\Gamma_{\alpha}$  at a vertex of  $\Gamma_{\alpha}$  is labeled with a vertex of  $\Gamma_{\beta}$ .

Two vertices of  $\Gamma_{\alpha}$  are *parallel* if the corresponding inner boundary components of  $P_{\alpha}$ , when given the orientation induced by some orientation of  $P_{\alpha}$ , are homologous in T. Otherwise, they are *antiparallel*.

Since all our manifolds are orientable, each arc in  $P_1 \cap P_2$  must join points of intersection of  $\partial P_1$  with  $\partial P_2$  of opposite sign. Thus we have the following *parity rule*:

If e is an interior edge of  $\Gamma_{\alpha}$  connecting parallel (resp. antiparallel) vertices of  $\Gamma_{\alpha}$ , then the labels at the endpoints of e represent antiparallel (resp. parallel) vertices of  $\Gamma_{\beta}$ .

Although in general the edges of our graphs are unoriented, it will sometimes be convenient to temporarily orient an edge e. Then  $\partial_+ e$  and  $\partial_- e$  will denote the head and tail endpoints of e respectively.

A cycle in  $\Gamma_{\alpha}$  is a subgraph homeomorphic to a circle. Equivalently, a cycle is a subgraph consisting of interior edges  $e_0, \ldots, e_{k-1}, k \ge 1$ , which may be oriented so that  $\partial_+ e_i = \partial_- e_{i+1}, i \in \mathbb{Z}_k$ , and so that the vertices  $\partial_+ e_i, i \in \mathbb{Z}_k$ , are distinct. We call k the length of the cycle.

Let x be a vertex of  $\Gamma_{\alpha}$ . An x-cycle in  $\Gamma_{\beta}$  is a cycle  $\sigma$  such that

(a) all the vertices of  $\sigma$  are parallel, and

(b) for some consistent orientation of the edges in  $\sigma$ , each edge e has label x at  $\partial_{-}e$ .

A Scharlemann cycle in  $\Gamma_{\beta}$  is an x-cycle  $\sigma$  (for some vertex x of  $\Gamma_{\alpha}$ ) such that the interior of the disk in  $D_{\beta}$  bounded by  $\sigma$  is disjoint from  $\Gamma_{\beta}$ . (The terminology is explained by the remark preceding Lemma 2.5.2 below.)

It follows easily from the definition that if the edges of a Scharlemann cycle in  $\Gamma_{\beta}$  are oriented consistently, then each edge e has label x at  $\partial_{-}e$  and label  $\bar{x}$ at  $\partial_{+}e$ , where x and  $\bar{x}$  are antiparallel and represent inner boundary components of  $P_{\alpha}$  which are adjacent on T. (See Figure 2.1.)



FIGURE 2.1

A Scharlemann cycle of length 1 will be called a trivial loop.

The next three lemmas are concerned with the topological interpretation of certain properties of the graphs  $\Gamma_1$  and  $\Gamma_2$ .

LEMMA 2.5.1.  $\Gamma_{\alpha}$  contains no trivial loop.

*Proof.* A trivial loop in  $\Gamma_{\alpha}$  gives rise to a boundary-compression of  $P_{\beta}$  towards T. It is easy to show that this implies that  $P_{\beta}$  is compressible.

Parts (a) and (b) of the next lemma are essentially contained in [Sch2, proof of Proposition 4.7] and [Sch3, proof of Proposition 5.6] respectively. We include the proofs for the convenience of the reader.

LEMMA 2.5.2. Suppose that  $\Gamma_{\beta}$  contains a Scharlemann cycle. Then:

(a) There exists a compressing disk  $D'_{\alpha}$  for S in  $M(r_{\alpha})$  with  $\partial D'_{\alpha} = \partial D_{\alpha}$  such that the corresponding planar surface  $P'_{\alpha}$  in M has two fewer inner boundary components than  $P_{\alpha}$ ; and

(b)  $M(r_{\alpha})$  has a lens space as a connected summand.

*Proof.* Let  $\sigma$  be a Scharlemann cycle in  $\Gamma_{\beta}$ . Then the length of  $\sigma$ , say k, is greater than 1 by Lemma 2.5.1.

(a)  $\sigma$  gives rise to a disk  $E \subset P_{\beta}$  whose boundary is the union of consecutive arcs  $a_0, b_0, \ldots, a_{k-1}, b_{k-1}$ , where the  $a_i$  are in  $P_1 \cap P_2$  and correspond to the edges of  $\sigma$ , and the  $b_i$  are in the inner boundary components of  $P_{\beta}$  corresponding to the vertices of  $\sigma$ . (See Figure 2.2.) Note that int  $E \cap P_{\alpha} = \emptyset$ . The labels at the endpoints of the edges of  $\sigma$  represent inner boundary components components of  $P_{\alpha}$  which are adjacent on T. These boundary components cobound an annulus A in T containing all the  $b_i$ . Let  $Q_{\alpha}$  be the punctured torus  $P_{\alpha} \cup A$ , pushed slightly into int M, and let  $P'_{\alpha}$  be  $Q_{\alpha}$  surgered along E. Since  $\partial E$  has algebraic (and geometric) intersection number k with a core of A,  $P'_{\alpha}$  is a planar surface with  $\partial_0 P'_{\alpha} = \partial_0 P_{\alpha}$  and with two fewer inner boundary components than  $P_{\alpha}$ .



FIGURE 2.2

(b) Consider the disk  $D_{\alpha} \subset M(r_{\alpha}) = M \cup J_{\alpha}$ . The inner boundary components of  $P_{\alpha}$  represented by the labels at the endpoints of the edges of  $\sigma$  bound meridian disks  $D, \overline{D}$ , say, in  $J_{\alpha}$ . Since these boundary components are antiparallel in T, the union of the 3-cell in  $J_{\alpha}$  bounded by  $D \cup A \cup \overline{D}$  with a suitable regular neighborhood of  $D_{\alpha}$  (pushed slightly off S) will be a solid torus V in int  $M(r_{\alpha})$ . Then  $\partial E \subset \partial V$ , and there is a regular neighborhood N(E) of E such

that  $V \cup N(E)$  is a punctured lens space in  $M(r_{\alpha})$  whose fundamental group has order k.

Remark. Usually we may assume without loss of generality that the disk  $D_{\alpha}$  is chosen so that the number  $n_{\alpha}$  of inner boundary components of  $P_{\alpha}$  is minimal. Then by Lemma 2.5.2(a), the existence of a Scharlemann cycle in  $\Gamma_{\beta}$  simply leads to a contradiction. One exception, however, occurs in the final step of the proof of Theorem 2.0.3. There the set-up is more complicated, and in particular it is not clear that we may assume that the number of inner boundary components of  $P_1$  is minimal while still maintaining the other hypotheses. In that case we use Lemma 2.5.2(b) instead.

Two edges e, e' of  $\Gamma_{\alpha}$  are *parallel* if the corresponding arcs a, a' in  $P_1 \cap P_2$  are parallel in  $P_{\alpha}$ , that is, there exists a disk E in  $P_{\alpha}$  such that  $\partial E = a \cup b \cup a' \cup b'$ , where b, b' are arcs in  $\partial P_{\alpha}$ . (Note that  $E \cap P_{\beta}$  may contain arcs other than a and a'.)

The next lemma is essentially contained in [Sho]. Recall that  $\Delta = \Delta(r_1, r_2)$ .

LEMMA 2.5.3. Suppose that  $P_{\alpha}$  is an annulus. Then either  $\Delta \leq 1$  or M is homeomorphic to  $T \times I$ .

Proof. Since  $P_{\alpha}$  is an annulus, all the edges of  $\Gamma_{\alpha}$ , and hence of  $\Gamma_{\beta}$ , are boundary edges (by Lemma 2.5.1). Suppose that  $\Delta \geq 2$ . Then by a simple outermost arc argument, there is a vertex y of  $\Gamma_{\beta}$  such that two adjacent (boundary) edges of  $\Gamma_{\beta}$  containing y are parallel. Let the corresponding arcs in  $P_1 \cap P_2$  be a and a'. Then, for i = 1, 2, there are arcs  $d_i \subset \partial D_i$ ,  $b_i \subset \partial P_i \partial D_i$ , and disks  $E_i \subset P_i$  such that  $\partial E_i = a \cup b_i \cup a' \cup d_i$ . Note that  $E_1 \cap E_2$  $-a \cup a'$  consists of arcs which are parallel into both  $d_1$  and  $d_2$ . These are easily removed, for example by an isotopy of  $E_2$  (using the incompressibility of S in M and the irreducibility of M). Then  $A = E_1 \cup E_2$  is an annulus in M with one boundary component in S and the other,  $b_1 \cup b_2 = \partial_0 A$ , say, in T. Since int  $b_{\beta} \cap \partial P_{\alpha} = \emptyset$ , we see that  $\partial_0 A$  has slope  $r_0$ , say, where  $\Delta(r_0, r_{\alpha}) = 1$  (see Figure 2.3). Hence, after an isotopy of  $A, A \cap P_{\alpha}$  will consist of a single spanning arc. Thus  $A \cup P_{\alpha}$  is homeomorphic to  $X \times I$ , where  $X = X \times \{0\}$  is



FIGURE 2.3

the union of two simple loops in T which intersect transversely in a single point, and has a regular neighborhood homeomorphic to  $N \times I$  where  $N = N \times \{0\}$  is a regular neighborhood of X in T. Since  $\partial N = \partial N \times \{0\}$  bounds a disk  $B_0$  in Tand S is incompressible in M,  $\partial N \times \{1\}$  bounds a disk  $B_1$  in S. Then  $B_0 \cup \partial N \times I \cup B_1$  is a 2-sphere, which bounds a 3-ball B in M since M is irreducible. Then  $N \times I \cup B \cong T \times I$ , showing that  $M \cong T \times I$ .

LEMMA 2.5.4. Suppose that  $\Gamma_{\alpha}$  contains boundary edges which are parallel in both  $\Gamma_1$  and  $\Gamma_2$ . Then either M contains an annulus with one boundary component in S and the other having slope  $r_0$  in T, where  $\Delta(r_0, r_i) = 1$ , i = 1, 2, or M is homeomorphic to  $T \times I$ .

*Proof.* Let the arcs in  $P_1 \cap P_2$  corresponding to the edges that are parallel in  $\Gamma_1$  and  $\Gamma_2$  be a and a'. Then, for i = 1, 2, there are arcs  $d_i \subset \partial D_i$ ,  $b_i \subset \partial P_i - \partial D_i$ , and disks  $E_i \subset P_i$ , such that  $\partial E_i = a \cup b_i \cup a' \cup d_i$ . By possibly rechoosing a' and performing an isotopy of  $E_2$  (say), we may assume that int  $E_1 \cap$  int  $E_2 = \emptyset$ . Then  $A = E_1 \cup E_2$  is an annulus in M with one boundary component in S and the other,  $b_1 \cup b_2 = \partial_0 A$ , say, in T. By moving  $\partial_0 A$ slightly into general position with respect to the component of  $\partial P_i$  containing  $b_i$ , we see that  $\partial_0 A$  has slope  $r_0$  in T where  $\Delta(r_0, r_i) > 0$ , i = 1, 2. Now apply Lemma 2.5.3 with A replacing  $P_{\alpha}$  in the statement of that lemma. We conclude that either  $M \cong T \times I$  or  $\Delta(r_0, r_i) = 1$ , i = 1, 2.

We shall usually use Lemma 2.5.4 in conjunction with the following observation.

LEMMA 2.5.5. Suppose that  $\Gamma_{\alpha}$  contains a family of  $2n_{\beta}$  mutually parallel boundary edges. Then  $\Gamma_{\alpha}$  contains boundary edges which are parallel in both  $\Gamma_1$  and  $\Gamma_2$ .

Proof. The edges in the parallel family may be numbered  $e(1), \ldots, e(n_{\beta}), e'(1), \ldots, e'(n_{\beta})$  so that e(y) and e'(y) have label y (at their vertex endpoint). Thus e(y) and e'(y) correspond to boundary edges in  $\Gamma_{\beta}$  with vertex endpoint y; let the union of these boundary edges be f(y). Then f(y) is a properly embedded arc in  $D_{\beta}$  containing the vertex y. Let  $y_0$  be a vertex such that  $f(y_0)$  is an outermost such arc. Then the edges  $e(y_0)$  and  $e'(y_0)$  are parallel in both  $\Gamma_1$  and  $\Gamma_2$ .

We now state the main graph-theoretic propositions on which the proof of the theorems in Section 2.4 and the proof of conclusion (iii) of Theorem 2.0.3 will be based. These propositions will be proved in Section 2.6. **PROPOSITION 2.5.6.** Suppose that  $\Delta \geq 2$ . Then either

(i)  $\Gamma_1$  or  $\Gamma_2$  contains a Scharlemann cycle; or

(ii) every vertex of  $\Gamma_{\alpha}$  belongs to a boundary edge of  $\Gamma_{\alpha}$ .

Addendum 2.5.7. In conclusion (ii) of Proposition 2.5.6, every vertex x of  $\Gamma_{\alpha}$  belongs to a boundary edge e(x) of  $\Gamma_{\alpha}$  such that the vertices of  $\Gamma_{\beta}$  in the set { label of e(x) at x: x a vertex of  $\Gamma_{\alpha}$  } are all parallel.

**PROPOSITION 2.5.8.** Suppose that  $\Delta \geq 3$ . Then either

(i)  $\Gamma_1$  or  $\Gamma_2$  contains a Scharlemann cycle; or

(ii) there exists a pair of boundary edges which are parallel in both  $\Gamma_1$  and  $\Gamma_2$ .

To complete the proof of Theorem 2.0.3 we need to make use of some additional structure. Thus, for the next proposition we assume that there is an essential surface  $(F, \partial F) \subset (M, T)$  with  $\partial F \neq \emptyset$  and having boundary slope  $r_1$ , and that the compressing disk  $D_1$  for S in  $M(r_1)$  is disjoint from F. We may assume that  $P_1$  is incompressible in M - F, for example, by choosing  $D_1$  so that  $n_1$  is minimal among all compressing disks for S in M - F. Since F is incompressible in M, this implies that  $P_1$  is also incompressible in M.

**PROPOSITION 2.5.9.** Suppose that  $\Delta \geq 2$ . Then either

(i)  $\Gamma_1$  contains a Scharlemann cycle; or

(ii)  $\Gamma_2$  contains a Scharlemann cycle; or

(iii) all the inner boundary components of  $P_1$  lie in a single component of  $T - \partial F$ .

This proposition will be used to establish conclusion (iii) of Theorem 2.0.3. Its proof is a modification of the proof of Proposition 2.5.8.

In the remainder of this section we show how Theorems 2.4.2–2.4.6, and the rest of Theorem 2.0.3 (together with Addendum 2.0.4), follow from the above graph-theoretic propositions.

**Proof of Theorem 2.4.2.** We may assume that  $n_{\alpha}$  is minimal over all compressing disks for S in  $M(r_{\alpha})$ . The theorem then follows immediately from Proposition 2.5.8, Lemma 2.5.2(a), and Lemma 2.5.4.

Proof of Theorem 2.4.3. By hypothesis, M contains an annulus  $A_0$  with one boundary component in S and the other having slope  $r_0$  in T.

(a)  $A_0$  gives rise to a compressing disk for S in  $M(r_0)$ .

(b) This follows immediately from Lemma 2.5.3.

(c) Let  $t: M \to M$  be the homeomorphism defined by Dehn twisting (in some direction) along  $A_0$ . If  $\Delta(r_i, r_0) = 1$ , i = 1, 2, then, for some integer k,

283

 $t^k$  takes the slope  $r_1$  to  $r_2$ . It therefore extends to a homeomorphism from  $M(r_1)$  to  $M(r_2)$ .

Proof of Theorem 2.4.4. Suppose that  $\Delta \geq 2$ , and that  $D_{\alpha}$  is chosen so that  $n_{\alpha}$  is minimal. By Proposition 2.5.6, Addendum 2.5.7, and Lemma 2.5.2(a), each vertex x of  $\Gamma_{\alpha}$  is an endpoint of a boundary edge e(x) of  $\Gamma_{\alpha}$  such that the labels of the edges e(x) at the corresponding vertices x represent parallel vertices of  $\Gamma_{\beta}$ . Since S is a torus, we may also assume that all the intersections of  $\partial D_1$  with  $\partial D_2$  have the same sign. By the same reasoning as that which led to the parity rule, these together imply that the vertices of  $\Gamma_{\alpha}$  are all parallel; similarly, the vertices of  $\Gamma_{\beta}$  are all parallel. By the parity rule, this implies that  $\Gamma_{\alpha}$  contains no interior edges. By a straightforward outermost arc argument, there exists a vertex x of  $\Gamma_{\alpha}$  such that all the (boundary) edges of  $\Gamma_{\alpha}$  incident to x are parallel. Since  $\Delta \geq 2$ , Lemmas 2.5.4 and 2.5.5 imply that either M is homeomorphic to  $T \times I$  or there is an annulus A in M with one boundary component, say  $\partial_1 A$ , in S, and the other, say  $\partial_0 A$ , having slope  $r_0$  in T, where  $\Delta(r_0, r_i) = 1$ , i = 1, 2.

Let K be a core of the solid torus J, where  $M(r_1) = M \cup J$ . Since  $\Delta(r_1, r_0) = 1$ , a meridian disk of J meets  $\partial_0 A$  in one point. Hence A can be extended to an annulus B such that  $\partial B = \partial_1 A \cup K$ , which can be used to isotope K into S. Now let  $\Sigma$  be a 2-sphere in  $M(r_1)$ . By the previous sentence, an isotopy of  $\Sigma$  in  $M(r_1)$  will move it off K, and hence into M, where it bounds a 3-ball. This shows that  $M(r_1)$  is irreducible. Therefore, since S compresses in  $M(r_1)$  by hypothesis,  $M(r_1)$  is a solid torus with boundary S. Thus M is the complement of an open regular neighborhood of a curve, K, in a solid torus which can be isotoped to an essential curve in the boundary of the solid torus. It follows that M is either a cable space or homeomorphic to  $T \times I$ .

Proof of Theorem 2.4.5. Let  $D_{\alpha}$  be a compressing disk for  $S_{\alpha}$  in  $M(r_{\alpha})$  with  $n_{\alpha}$  minimal. Since  $S_1 \cap S_2 = \emptyset$ , the graph  $\Gamma_{\alpha}$  contains no boundary edges. The result now follows from Proposition 2.5.6 and Lemma 2.5.2(a).

Proof of Theorem 2.4.6. Here the arcs in  $P_1 \cap P_2$  give rise to graphs  $\Gamma_1$ ,  $\Gamma_2$  in 2-spheres  $\Sigma_1$ ,  $\Sigma_2$ . All the remarks made at the beginning of this section apply, *mutatis mutandis*, to  $\Gamma_1$  and  $\Gamma_2$ . Hence, formally removing an open disk from  $\Sigma_i - \Gamma_i$ , i = 1, 2, we obtain two graphs in disks with no boundary edges, to which we may apply Proposition 2.5.6. We conclude that either  $\Delta(r_1, r_2) \leq 1$  or there is a Scharlemann cycle in  $\Gamma_1$  or  $\Gamma_2$ . It is easy to see that Lemma 2.5.1 still applies, since  $P_1$  and  $P_2$  are essential in M, so that any Scharlemann cycle in  $\Gamma_1$  or  $\Gamma_2$  has length greater than 1. But if  $\Gamma_\beta$  contains such a Scharlemann cycle, then an argument entirely analogous to the proof of Lemma 2.5.2(b) shows that  $M(r_\alpha)$  contains a lens space as a connected summand.

*Remark*. Although a Scharlemann cycle in  $\Gamma_{\beta}$  can also be used to construct a new planar surface  $P'_{\alpha}$  with two fewer boundary components, as in Lemma 2.5.2(a), there is now no guarantee that  $P'_{\alpha}$  is essential.

Proof of Theorem 2.0.3. There are two cases.

(1) F is non-planar. By Proposition 2.2.1, either conclusion (i) of Theorem 2.0.3 holds or M contains a closed incompressible surface S which is disjoint from F. Moreover, S has the additional properties described in Addendum 2.2.2.

If condition (i) of Addendum 2.2.2 holds, then S remains incompressible in M(s) whenever  $\Delta(r, s) > 1$  by Theorem 2.4.3(b). (Note that S is not parallel to  $\partial M$ , for example since it is disjoint from F.) This is conclusion (iii) of Theorem 2.0.3.

So suppose that S satisfies condition (ii) of Addendum 2.2.2. Write  $r_1 = r$ and let  $D_1$  be a compressing disk for S in  $N(r_1)$  which misses  $\hat{F}$ , as guaranteed by (ii)(b) of Addendum 2.2.2. By the remark immediately preceding the statement of Proposition 2.5.9, we may assume that the corresponding punctured disk  $P_1$  is incompressible in M. Suppose that S also compresses in  $M(r_2)$ , and therefore in  $N(r_2)$ , for some slope  $r_2$ . Let  $D_2$  be a compressing disk for S in  $N(r_2)$ , such that the number of inner boundary components of the corresponding planar surface  $P_2$  is minimal. Let  $\Gamma_1$ ,  $\Gamma_2$  be the graphs in  $D_1$ ,  $D_2$  obtained in the usual way. Now apply Proposition 2.5.9. Conclusion (i) of that proposition contradicts the minimality of the number of inner boundary components of  $P_2$ , by Lemma 2.5.2(a). Conclusion (ii) contradicts condition (ii)(a) of Addendum 2.2.2, in view of Lemma 2.5.2(b). Finally, conclusion (iii) contradicts condition (ii)(c) of Addendum 2.2.2. We must therefore have  $\Delta(r_1, r_2) \leq 1$ , and thus obtain conclusion (iii) of Theorem 2.0.3.

We point out that in the above proof we are not able to use the existence of a Scharlemann cycle in  $\Gamma_2$  to reduce the number of inner boundary components of  $P_1$ , as this might destroy the condition  $P_1 \cap F = \emptyset$  which is crucial for the proof of Proposition 2.5.9.

(2) *F* is planar. First suppose that *F* is connected. Then by Proposition 2.3.1, either conclusion (ii) of Theorem 2.0.3 holds or *M* contains an incompressible torus *S* which is disjoint from *F* and compresses in M(r). Therefore, by Theorem 2.0.1, either *S* remains incompressible in M(s) if  $\Delta(r, s) > 1$  (and we have conclusion (iii) of Theorem 2.0.3), or *S* and  $\partial M$  cobound a cable space. If the latter holds, then in particular there is an annulus *A* in *M* with one boundary component in *S* and the other having slope  $r_0$ , say, in *T*. Now consider  $A \cap F$ . Since  $F \cap S = \emptyset$  and *F* is essential in *M*, a standard outermost arc argument shows that we must have  $r_0 = r$ . Then again we obtain conclusion (iii) of Theorem 2.4.3(b).

If F is disconnected then, by Proposition 2.3.1(iii), M fibers over  $S^1$  with fiber G, and we have conclusion (iv) of Theorem 2.0.3.

Before giving the proof of Addendum 2.0.4, we need the following lemma.

LEMMA 2.5.10. Let M be a 3-manifold which fibers over  $S^1$  with fiber a connected surface, and let F be an essential connected surface in M. If either

(a) F is disjoint from some fiber; or

(b) F is planar and  $\partial F$  is disjoint from some fiber,

then F is isotopic to a fiber.

**Proof.** (a) Let H be a fiber such that  $F \cap H = \emptyset$ . Then F is a connected incompressible surface in  $H \times I$  (that is, M cut along H) such that  $F \cap H \times \partial I = \emptyset$ . But it follows easily from [Wa1, Proposition 3.1] that any such surface is either a boundary-parallel annulus or parallel to  $H \times \{0\}$ .

(b) Move F into general position with respect to some fiber H, keeping  $\partial F \cap H = \emptyset$ . Thus  $F \cap H$  consists of disjoint simple loops; let C be one that is innermost on F (which is planar by hypothesis), and let  $F_0$  be the corresponding innermost component of F cut along C. Then, when we cut M along H,  $F_0$  is an incompressible surface in  $H \times I$  with  $\partial F_0$  contained in (say)  $H \times \{0\} \cup \partial H \times I$  and  $\partial F_0 \cap H \times \{0\} \neq \emptyset$ . Again by [Wa1, Proposition 3.1] any such surface is parallel to a subsurface of  $H \times \{0\}$ , and so we may perform an isotopy of F to reduce the number of components of  $F \cap H$ . Continuing in this way, we eventually move F off H, at which point we apply part (a).

Proof of Addendum 2.0.4. Let P be a fiber. We work with a surface satisfying the following slight modification of the definition given in Section 2.1. Let  $(F, \partial F) \subset (M, \partial M)$  be an essential separating surface with boundary slope r, such that each component of F has non-empty boundary and is not isotopic to P, and such that the number of components of  $\partial F$  is minimal subject to these conditions. Since r is a strict boundary slope by hypothesis, there exists such a surface F. In addition, we assume (as we may) that if F is disconnected then it consists of two parallel copies of some connected non-separating surface. Note that by Lemma 2.5.10(b), F is not planar.

Recall the definition of  $F_i^-$  from Section 2.1. Our minimality assumption on F here implies that each component of  $F_i^-$  is either closed, or a boundary-parallel annulus, or non-separating, and that if there are any non-separating components, then at least one of them is isotopic to P. On the other hand, if  $F_i^-$  did have a component isotopic to P, then, since by a small isotopy we can make  $F \cap F_i^- = \emptyset$ , F would consist of copies of P by Lemma 2.5.10(a). We conclude that  $F_i^-$  consists of boundary-parallel annuli together with a closed surface.

The proofs of Proposition 2.2.1 and Addendum 2.2.2 then apply verbatim to this surface F. Since M(r) is homeomorphic to  $S^1 \times S^2$ , conclusion (i) of

Proposition 2.2.1 does not hold; so we have conclusion (ii), together with Addendum 2.2.2. The proof of the relevant part of Theorem 2.0.3 (case (1) above) now applies to give conclusion (iii) of that theorem.  $\Box$ 

#### 2.6. The combinatorics of graphs in disks

This section is graph-theoretic in nature. In it we assume that we are given graphs  $\Gamma_1$ ,  $\Gamma_2$  in disks  $D_1$ ,  $D_2$  as described in Section 2.5, and our goal is to prove Propositions 2.5.6 (including Addendum 2.5.7), 2.5.8, and 2.5.9.

The search for Scharlemann cycles is facilitated by consideration of the following more general type of cycle (see Lemma 2.6.2 below).

A great cycle (specifically, a great x-cycle) in  $\Gamma_{\beta}$  is an x-cycle  $\sigma$  such that all the vertices of  $\Gamma_{\beta}$  that lie in the (closed) disk in  $D_{\beta}$  bounded by  $\sigma$  are parallel.

Let  $\Lambda$  be a subgraph of  $\Gamma_{\beta}$ , and let x be a vertex of  $\Gamma_{\alpha}$ . We say that  $\Lambda$  satisfies condition P(x) if:

For each vertex y of  $\Lambda$  there exists an edge of  $\Lambda$  incident to y with label x, connecting y to a parallel vertex of  $\Lambda$ .

LEMMA 2.6.1. Suppose that  $\Lambda$  satisfies condition P(x). Then each component of  $\Lambda$  contains an x-cycle.

*Proof.* For each vertex y of  $\Lambda$ , choose an edge e(y) of  $\Lambda$  incident to y with label x, connecting y to a parallel vertex of  $\Lambda$ .

Let  $\Lambda_0$  be a component of  $\Lambda$ , and let  $y_1$  be any vertex of  $\Lambda_0$ . Consider the edge  $e(y_1)$ , connecting  $y_1$  to  $y_2$ , say. Since  $y_1$  and  $y_2$  are parallel, the label of  $e(y_1)$  at  $y_2$  is not x, by the parity rule. Hence if  $y_2 \neq y_1$ , the edge  $e(y_2)$  is distinct from  $e(y_1)$ . Continue in this way, obtaining edges  $e(y_i)$  connecting vertices  $y_i$  and  $y_{i+1}$ , with label x at  $y_i$ , until a vertex is repeated for the first time, say  $y_m = y_n$ , m < n, but  $y_i \neq y_j$  for  $1 \le i < j < n$ . Then the edges  $e(y_m), \ldots, e(y_{n-1})$  form an x-cycle in  $\Lambda_0$ .

LEMMA 2.6.2. If  $\Gamma_{\beta}$  contains a great cycle, then it contains a Scharlemann cycle.

*Proof.* Let  $\sigma$  be a great x-cycle in  $\Gamma_{\beta}$ , for some vertex x of  $\Gamma_{\alpha}$ , and let E be the disk in  $D_{\beta}$  bounded by  $\sigma$ . Let  $\epsilon(\sigma)$  be the number of edges e in  $\Gamma_{\beta}$  such that int  $e \subset$  int E. We prove the result by induction on  $\epsilon(\sigma)$ .

If  $\epsilon(\sigma) = 0$ , then  $\sigma$  is a Scharlemann cycle by definition. So suppose that  $\epsilon(\sigma) > 0$ . We distinguish two cases:

(1) Any edge in  $\Gamma_{\beta} \cap E$  incident to a vertex of  $\sigma$  lies in  $\sigma$ .

Since all the vertices of  $\Gamma_{\beta} \cap E$  are parallel, the graph  $\Lambda = \Gamma_{\beta} \cap E - \sigma$ satisfies P(x') for any vertex x' of  $\Gamma_{\alpha}$ , and hence contains an x'-cycle  $\sigma'$  by Lemma 2.6.1. Clearly  $\sigma'$  is a great x'-cycle in  $\Gamma_{\beta}$  and  $\epsilon(\sigma') < \epsilon(\sigma)$ . Hence the result follows by induction.

(2) There exists an edge of  $\Gamma_{\beta} \cap E$  incident to a vertex of  $\sigma$  but not contained in  $\sigma$ .

Since all the vertices of  $\sigma$  are parallel, there is a label x' such that x and x' represent inner boundary components of  $P_{\alpha}$  which are adjacent in T, with the property that for every vertex y of  $\sigma$  there is an edge of  $\Gamma_{\beta} \cap E$  incident to y with label x'. This is also (trivially) true for any vertex y in int E. Hence  $\Gamma_{\beta} \cap E$  satisfies condition P(x'). An application of Lemma 2.6.1 would now give an x'-cycle  $\sigma'$  in  $\Gamma_{\beta} \cap E$ ; however, we want to avoid the possibility that  $\sigma' = \sigma$ . To do this, note that since by hypothesis there exists an edge of  $\Gamma_{\beta} \cap E$  incident to a vertex g, say, in  $\sigma$  with label x'. Let e be the edge of  $\sigma$  which, when oriented so that it has label x at  $\partial_{-}e$ , has  $\partial_{+}e = g$ . Then the graph  $\Lambda = \Gamma_{\beta} \cap E -$  int e also satisfies condition P(x'). By Lemma 2.6.1 we obtain an x'-cycle  $\sigma'$  in  $\Lambda$ , which is clearly a great x'-cycle in  $\Gamma_{\beta}$  with  $\epsilon(\sigma') < \epsilon(\sigma)$ . Then  $\Gamma_{\beta}$  contains a Scharlemann cycle by induction.

Let  $\Lambda$  be a graph in a disk D, and choose a point  $\infty \in \partial D - \Lambda$ . We may then define a partial ordering on the set of components of  $\Lambda$  by declaring that  $\Lambda_1 < \Lambda_2$  if and only if every path in D from  $\Lambda_1$  to  $\infty$  meets  $\Lambda_2$ . Call a component of  $\Lambda$  extremal if it is minimal with respect to this partial ordering for some choice of  $\infty$ .

To prove Proposition 2.5.6 we focus on the following condition:

 $(*)_{\alpha}$ . There exists a vertex x of  $\Gamma_{\alpha}$  such that for each vertex y of  $\Gamma_{\beta}$  there is an edge of  $\Gamma_{\alpha}$  incident to x with label y, connecting x to an antiparallel vertex of  $\Gamma_{\alpha}$ .

LEMMA 2.6.3. Suppose that condition  $(*)_{\alpha}$  holds. Then  $\Gamma_{\beta}$  contains a great x-cycle.

**Proof.** By the parity rule, the hypothesis  $(*)_{\alpha}$  is equivalent to the condition that there exists a vertex x of  $\Gamma_{\alpha}$  such that for each vertex y of  $\Gamma_{\beta}$  there is an edge e(y) of  $\Gamma_{\beta}$  incident to y with label x, connecting y to a parallel vertex of  $\Gamma_{\beta}$ . For each vertex y of  $\Gamma_{\beta}$ , choose such an edge e(y), and let  $\Lambda$  be the subgraph of  $\Gamma_{\beta}$  consisting of all these edges e(y). Then  $\Lambda$  satisfies condition P(x). Note also that each component of  $\Lambda$  has all its vertices parallel.

Let  $\Lambda_0$  be an extremal component of  $\Lambda$ . By Lemma 2.6.1,  $\Lambda_0$  contains an *x*-cycle  $\sigma$ . Since  $\Lambda_0$  is extremal, all vertices of  $\Gamma_{\beta}$  in the disk bounded by  $\sigma$  belong to  $\Lambda_0$ ; in particular, they are parallel. Hence  $\sigma$  is a great *x*-cycle.

The negation of condition  $(*)_{\alpha}$  is:

 $(**)_{\alpha}$ . For each vertex x of  $\Gamma_{\alpha}$  there exists a vertex y(x) of  $\Gamma_{\beta}$  such that each edge of  $\Gamma_{\alpha}$  incident to x with label y(x) connects x either to a parallel vertex of  $\Gamma_{\alpha}$  or to  $\partial D_{\alpha}$ .

LEMMA 2.6.4. Suppose that  $\Delta \geq 2$  and that condition  $(**)_{\alpha}$  holds. Then either

(i)  $\Gamma_{\alpha}$  contains a great cycle; or

(ii) every vertex of  $\Gamma_{\beta}$  belongs to a boundary edge of  $\Gamma_{\beta}$ .

**Proof.** For each vertex x of  $\Gamma_{\alpha}$  choose a vertex y(x) of  $\Gamma_{\beta}$  as in condition  $(**)_{\alpha}$ , and define the subgraph  $\Lambda$  of  $\Gamma_{\alpha}$  to be the union over all vertices x of  $\Gamma_{\alpha}$  of {edges of  $\Gamma_{\alpha}$  incident to x with label y(x)}. Note that each vertex of  $\Lambda$  has valency  $\geq \Delta \geq 2$ . Also, each component of  $\Lambda$  has all its vertices parallel.

Let  $\Lambda_0$  be an extremal component of  $\Lambda$  with respect to some point  $\infty \in \partial D_{\alpha} - \Lambda$ . Let R be the component of  $D_{\alpha} - \Lambda_0$  containing  $\infty$ . The frontier FrR of R can be expressed as the union of a sequence of oriented edges  $e_1, \ldots, e_n$  of  $\Lambda_0$ , with  $\partial_+ e_i = \partial_- e_{i+1}$ ,  $1 \le i \le n - 1$ , such that as we traverse these edges in order, the component of  $D_{\alpha} - \Lambda_0$  immediately on (say) our left is always R. (As oriented edges,  $e_1, \ldots, e_n$  are distinct, although as unoriented edges there may be repetitions. Note however that, since each vertex of  $\Lambda_0$  has valency at least 2, consecutive edges in the sequence are distinct as unoriented edges.)

Recall from Section 2.5 the definition of a cycle. We distinguish two cases: (1) The sequence  $e_1, \ldots, e_n$  contains a cycle.

Note that this is necessarily the case if  $\Lambda_0 \subset \operatorname{int} D_{\alpha}$ .

Let the cycle be  $\sigma$ , with edges  $f_0 = e_m, \ldots, f_{k-1} = e_{m+k-1}$ , say; so  $\partial_+ f_i = \partial_- f_{i+1}$ ,  $i \in \mathbb{Z}_k$ . Let  $x_i$  be the vertex  $\partial_- f_i$ ,  $i \in \mathbb{Z}_k$ . Let E be the disk in  $D_{\alpha}$  bounded by  $\sigma$ . (See Figure 2.4.)



FIGURE 2.4

Observe that, except possibly for i = 0, as we encircle the vertex  $x_i$  in the clockwise direction, the edges of  $\Gamma_{\alpha}$  incident to  $x_i$  between  $f_{i-1}$  and  $f_i$  have their interiors in R, while the remaining edges of  $\Gamma_{\alpha}$  incident to  $x_i$  lie in E. (See Figure 2.5.) Furthermore, since by the definition of  $\Lambda$  there are edges of  $\Lambda_0$  incident to  $x_i$  at least at each of the  $\Delta$  occurrences of the label  $y(x_i)$ , there are at most  $n_{\beta} - 1$  incidences of edges in the interior of R between  $f_{i-1}$  and  $f_i$ . It follows that if  $i \neq 0$ , then for each vertex y of  $\Gamma_{\beta}$  there is an edge of  $\Gamma_{\alpha} \cap E$  incident to  $x_i$  with label y. This is also (trivially) true for any vertex in int E.



FIGURE 2.5

Since  $\Lambda_0$  is extremal, all vertices of  $\Gamma_{\alpha} \cap E$  are in  $\Lambda_0$ , and hence are parallel. Let  $y_0$  be the label at  $x_0$  of any edge in  $\Gamma_{\alpha} \cap E$  incident to  $x_0$  (for example,  $f_0$ ). Then  $\Gamma_{\alpha} \cap E$  satisfies condition  $P(y_0)$ . By Lemma 2.6.1,  $\Gamma_{\alpha} \cap E$  contains a  $y_0$ -cycle. Since all vertices of  $\Gamma_{\alpha} \cap E$  are parallel, this is necessarily a great  $y_0$ -cycle, giving conclusion (i).

(2) The sequence  $e_1, \ldots, e_n$  does not contain a cycle.

Then  $\operatorname{Fr} R$  is an arc in  $\Lambda_0$  with its endpoints in  $\partial D_{\alpha}$ . Let E be the disk  $D_{\alpha} - R$ . (See Figure 2.6.) As in case (1) above, for every vertex x in  $\operatorname{Fr} R$  and every vertex y of  $\Gamma_{\beta}$  there is an edge of  $\Gamma_{\alpha} \cap E$  incident to x with label y.

Suppose that conclusion (ii) does not hold, so that there is a vertex  $y_0$  of  $\Gamma_{\beta}$  which does not lie in any boundary edge of  $\Gamma_{\beta}$ . Then any edge of  $\Gamma_{\alpha}$  incident to



FIGURE 2.6

any vertex with label  $y_0$  is an interior edge. Since  $\Lambda_0$  is extremal, all vertices of  $\Gamma_{\alpha} \cap E$  are in  $\Lambda_0$ , and hence are parallel. Therefore  $\Gamma_{\alpha} \cap E$  satisfies condition  $P(y_0)$ . By Lemma 2.6.1,  $\Gamma_{\alpha} \cap E$  contains a  $y_0$ -cycle, which is necessarily a great  $y_0$ -cycle since all vertices of  $\Gamma_{\alpha} \cap E$  are parallel. Thus we have conclusion (i).  $\Box$ 

Remark. In the last paragraph of the above proof, we only used the fact that there was a vertex  $y_0$  of  $\Gamma_{\beta}$  such that any edge of  $\Gamma_{\alpha} \cap E$  (as opposed to any edge of  $\Gamma_{\alpha}$ ) incident to any vertex with label  $y_0$  was an interior edge. Since all the vertices of  $\Gamma_{\alpha} \cap E$  are parallel, conclusion (ii) of Lemma 2.6.4 may therefore be strengthened to say that every vertex y of  $\Gamma_{\beta}$  is an endpoint of a boundary edge e(y) of  $\Gamma_{\beta}$  such that the vertices of  $\Gamma_{\alpha}$  in the set {label of e(y) at y: y a vertex of  $\Gamma_{\beta}$ } are all parallel.

We can now prove Proposition 2.5.6 and Addendum 2.5.7.

*Proof of Proposition* 2.5.6. This follows immediately from Lemmas 2.6.2, 2.6.3 and 2.6.4.  $\hfill \Box$ 

Proof of Addendum 2.5.7. This follows from the remark after the proof of Lemma 2.6.4.  $\hfill \Box$ 

The following lemma will be needed in the proofs of Propositions 2.5.8 and 2.5.9.

LEMMA 2.6.5. Let  $\Gamma$  be a graph in a disk D with no trivial loops or parallel edges, such that every vertex of  $\Gamma$  belongs to a boundary edge. Then  $\Gamma$  has a vertex of valency at most 3 which belongs to a single boundary edge.

Thus  $\Gamma$  has a vertex of one of the types illustrated in Figure 2.7 (which shows all the edges of  $\Gamma$  incident to the vertex).



FIGURE 2.7

**Proof.** For the purposes of this proof only, let us call vertices x, x' of  $\Gamma$  adjacent if there exist edges d, d' of  $\Gamma$  connecting x, x' to  $\partial D$  such that  $d \cap \partial D$  and  $d' \cap \partial D$  are adjacent on  $\partial D$  among all points of  $\Gamma \cap \partial D$ .

Let the number of vertices of  $\Gamma$  be n. The lemma is clearly true if  $n \leq 3$ . For  $n \geq 4$  we will prove the following stronger assertion by induction on n: (\*) There exist two non-adjacent vertices of valency at most 3, each belonging to a single boundary edge.

So suppose  $n \ge 4$  and assume that (\*) is true for graphs with m vertices,  $4 \le m < n$ . We consider three cases:

(1) There is an edge of  $\Gamma$  connecting non-adjacent vertices.

Let e be such an edge connecting x and x', say. Let d, d' be edges connecting x, x' to  $\partial D$ . (See Figure 2.8.) Let  $D_1, D_2$  be the closures of the components of  $D - d \cup e \cup d'$ , and consider the graphs  $\Gamma_i = \Gamma \cap D_i$  in D, i = 1, 2. Suppose  $\Gamma_i$  has  $n_i$  vertices. Since x, x' are not adjacent in  $\Gamma$ , we must have  $n_i \geq 3$ , i = 1, 2, or equivalently  $n_i < n$ , i = 1, 2. Note that x, x' are adjacent in  $\Gamma_i$ . We claim that there exists a vertex  $y_i$  of  $\Gamma_i$  of valency at most 3 which belongs to a single boundary edge of  $\Gamma_i$ , with  $y_i \neq x$  or x'. If  $n_i \geq 4$ , this follows from the inductive hypothesis applied to  $\Gamma_i$ . If  $n_i = 3$ , then take  $y_i$  to be the vertex of  $\Gamma_i$  not equal to x or x'. Now  $y_1, y_2$  are non-adjacent vertices of  $\Gamma$  of valency at most 3, each belonging to a single boundary edge.



FIGURE 2.8

(2) There is a vertex of  $\Gamma$  which belongs to two distinct boundary edges.

Let x be such a vertex, contained in boundary edges d, d', say. (See Figure 2.9.) Let  $D_1, D_2$  be the closures of the components of  $D - d \cup d'$ , and consider



FIGURE 2.9

the graphs  $\Gamma_i = \Gamma \cap D_i$  in D, i = 1, 2. Suppose that  $\Gamma_i$  has  $n_i$  vertices. Since d and d' are not parallel in  $\Gamma$ , we must have  $n_i \ge 2$ , i = 1, 2, or equivalently  $n_i < n$ , i = 1, 2. We claim that there exists a vertex  $y_i$  of  $\Gamma_i$  of valency at most 3, which belongs to a single boundary edge of  $\Gamma_i$ , with  $y_i \ne x$ . If  $n_i \ge 4$ , this follows from the inductive hypothesis applied to the graph  $\overline{\Gamma}_i$  obtained from  $\Gamma_i$  by amalgamating the parallel edges d, d'. If  $n_i = 2$  or 3, then the claim is easily verified by inspection. Now  $y_1, y_2$  satisfy (\*) for  $\Gamma$ .

(3) Neither (1) nor (2) holds.

Then every vertex belongs to a single boundary edge, and is connected to at most two vertices (since they must be adjacent to it). Thus any two non-adjacent vertices satisfy (\*).  $\Box$ 

LEMMA 2.6.6. If  $\Gamma_{\alpha}$  contains a parallel family of edges connecting parallel vertices, then either the sets of labels at the two ends of the family are disjoint, or  $\Gamma_{\alpha}$  contains a Scharlemann cycle.

The following corollary is immediate.

COROLLARY 2.6.7. If  $\Gamma_{\alpha}$  contains a parallel family of more than  $n_{\beta}/2$  edges connecting parallel vertices, then  $\Gamma_{\alpha}$  contains a Scharlemann cycle.

Proof of Lemma 2.6.6. Assume that the sets of labels at the two ends of the parallel family are not disjoint, and let y be a vertex of  $\Gamma_{\beta}$  which appears in both sets. Since no edge in the family can have label y at both ends by the parity rule, we see that the family contains a y-cycle  $\sigma$  of length 2. Since there are no vertices in the interior of the disk that  $\sigma$  bounds,  $\sigma$  is a great y-cycle. Now apply Lemma 2.6.2 to conclude that  $\Gamma_{\alpha}$  contains a Scharlemann cycle.

LEMMA 2.6.8. If  $\Gamma_{\alpha}$  contains a parallel family of  $n_{\beta}$  edges connecting antiparallel vertices, then  $\Gamma_{\beta}$  contains a Scharlemann cycle.

*Proof.* This follows immediately from Lemmas 2.6.3 and 2.6.2.  $\Box$ 

Proof of Proposition 2.5.8. By Proposition 2.5.6 we may assume that each vertex of  $\Gamma_{\alpha}$  belongs to a boundary edge. Let  $\overline{\Gamma}_{\alpha}$  be the reduced graph corresponding to  $\Gamma_{\alpha}$ , obtained by amalgamating all mutually parallel edges in the obvious way. Applying Lemma 2.6.5 to  $\overline{\Gamma}_{\alpha}$ , we conclude that  $\Gamma_{\alpha}$  has a vertex x of one of the forms illustrated in Figure 2.10.



We treat these separately.

Type I. In this case  $\Gamma_{\alpha}$  contains  $\Delta n_{\beta} \geq 3n_{\beta}$  parallel boundary edges, giving a pair of boundary edges which are parallel in both  $\Gamma_1$  and  $\Gamma_2$  by Lemma 2.5.5.

Type II. Since  $\Delta \geq 3$ , there must be either at least  $n_{\beta}$  parallel edges connecting x to y, or at least  $2n_{\beta}$  parallel edges connecting x to  $\partial D_{\alpha}$ . In the first case, we obtain a Scharlemann cycle in  $\Gamma_1$  or  $\Gamma_2$  by Corollary 2.6.7 and Lemma 2.6.8, and in the second case we obtain a pair of boundary edges which are parallel in both  $\Gamma_1$  and  $\Gamma_2$  by Lemma 2.5.5.

Type III. Let U, V, W be the sets of labels around x corresponding to the edges connecting x to u, v and  $\partial D_{\alpha}$  respectively. (See Figure 2.11.)We denote by |U| the number of labels (counted with multiplicity) in U, etc. Then  $|U| + |V| + |W| = \Delta n_{\beta} \ge 3n_{\beta}$ . There are three sub-cases:

(1) u, v and x parallel. By Corollary 2.6.7, either  $\Gamma_{\alpha}$  contains a Scharlemann cycle or  $|U| \leq (1/2)n_{\beta}$ ,  $|V| \leq (1/2)n_{\beta}$ . But then  $|W| \geq 2n_{\beta}$ , giving a pair of boundary edges which are parallel in both  $\Gamma_1$  and  $\Gamma_2$ .



FIGURE 2.11

(2) u, v parallel, antiparallel to x. If  $|W| \ge 2n_{\beta}$ , we obtain a pair of boundary edges which are parallel in both  $\Gamma_1$  and  $\Gamma_2$ . If not, then  $|U| + |V| \ge n_{\beta}$ , which implies that every vertex of  $\Gamma_{\beta}$  occurs as the label at x of some edge of  $\Gamma_{\alpha}$  connecting x to either u or v. Then  $\Gamma_{\beta}$  contains a great x-cycle by Lemma 2.6.3, and consequently a Scharlemann cycle by Lemma 2.6.2.

(3) u, v antiparallel. We may assume without loss of generality that u is parallel to x.

Let  $A = \{ \text{vertices } y \text{ of } \Gamma_{\beta} : y \text{ appears at least twice as a label in } W \}$ , and  $B = \{ \text{vertices } y \text{ of } \Gamma_{\beta} : y \text{ appears as a label in both } U \text{ and } V \}$ . Since each vertex of  $\Gamma_{\beta}$  appears  $\Delta \geq 3$  times in  $U \cup V \cup W$ , and since by Corollary 2.6.7 and Lemma 2.6.8 we may assume that no vertex appears more than once in either U or V, it follows that every vertex of  $\Gamma_{\beta}$  belongs to either A or B.

If  $y \in A$ , then there exist two distinct edges  $e_1(y)$ ,  $e_2(y)$  in  $\Gamma_{\beta}$  connecting y to  $\partial D_{\beta}$  (with label x at y), such that  $e_1(y)$  and  $e_2(y)$  are parallel in  $\Gamma_{\alpha}$ .

If  $y \in B$ , then there exist edges  $e_u(y)$  (resp.  $e_v(y)$ ) in  $\Gamma_{\beta}$ , incident to y with label x, connecting y to an antiparallel (resp. parallel) vertex of  $\Gamma_{\beta}$ .

Let  $\Lambda$  be the subgraph of  $\Gamma_{\beta}$  consisting of  $\{e_1(y), e_2(y): y \in A\}$ . Then  $\Lambda$  is a disjoint union of properly embedded arcs in  $\Gamma_{\beta}$ , each containing a single vertex. Consider an extremal component  $\Lambda_0$  of  $\Lambda$ , containing the vertex  $y_0$ , say. Then (at least) one of the two components of  $D_{\beta} - \Lambda_0$ , call it R, contains no vertices of  $\Lambda$ . Thus all vertices of  $\Gamma_{\beta}$  in R belong to B.

We now distinguish two cases.

(3.A) R contains no vertices of  $\Gamma_{\beta}$ . Then the edges  $e_1(y)$  and  $e_2(y)$  are parallel in  $\Gamma_{\beta}$ . Since they are also parallel in  $\Gamma_{\alpha}$ , we are done.

(3.B) *R* contains a vertex of  $\Gamma_{\beta}$ . Every vertex of *R* belongs to *B*, and hence in particular is connected by edges of  $\Gamma_{\beta}$  to both a parallel and an antiparallel vertex of  $\Gamma_{\beta}$ . It follows that the vertices of  $\Gamma_{\beta}$  in *R* cannot all be parallel to  $y_0$ . Let  $\Pi$  be the subgraph of  $\Gamma_{\beta}$  defined by  $\Pi = \{e_v(y): y \in R \text{ and } y \text{ is}$ antiparallel to  $y_0\}$ . Then  $\Pi$  satisfies condition P(x), and hence contains an *x*-cycle  $\sigma$  by Lemma 2.6.1. By Proposition 2.5.6, we may assume that every vertex of  $\Gamma_{\beta}$  belongs to a boundary edge, which implies that there are no vertices of  $\Gamma_{\beta}$  in the interior of the disk bounded by  $\sigma$ . Hence  $\sigma$  is a great *x*-cycle, and so  $\Gamma_{\beta}$  contains a Scharlemann cycle by Lemma 2.6.2.

Proof of Proposition 2.5.9. Recall that, in addition to the usual set-up, there is an essential surface  $(F, \partial F) \subset (M, T)$  with  $\partial F \neq \emptyset$  and boundary slope  $r_1$ , such that  $P_1 \cap F = \emptyset$ . Consider  $P_2 \cap F$ . The arcs in  $P_2 \cap F$  define a graph  $\Gamma_0$ , say, in  $D_2$ , such that  $\Gamma_0 \cap \Gamma_2 = \{$ vertices of  $\Gamma_0 \} = \{$ vertices of  $\Gamma_2 \}$ . Since F is essential in M, it is easy to see that the proof of Lemma 2.5.1 allows us to assume that  $\Gamma_0$  contains no trivial loops. Also, since  $F \cap S = \emptyset$  and  $\partial D_2 \subset S$ ,  $\Gamma_0$  contains no boundary edges.

Claim 2.6.9.  $\Gamma_2$  contains no parallel family of more than  $n_1$  boundary edges.

*Proof.* As we go around an inner boundary component of  $P_2$ , between successive intersections with any given inner boundary component of  $P_1$ , we must encounter all the boundary components of F. Since  $\Gamma_0$  has no boundary edges and no trivial loops, the claim follows.

Continuing with the proof of Proposition 2.5.9, suppose that  $\Delta \ge 2$ . Exactly as in the first paragraph of the proof of Proposition 2.5.8, we conclude that if neither  $\Gamma_1$  nor  $\Gamma_2$  contains a Scharlemann cycle, then  $\Gamma_2$  has a vertex x of one of the types illustrated in Figure 2.10.

Type I. Then  $\Gamma_2$  contains a parallel family of  $\Delta n_1 \ge 2n_1$  boundary edges, contradicting Claim 2.6.9.

*Type* II. Since there are at most  $n_1$  parallel edges connecting x to  $\partial D_2$  by Claim 2.6.9, there are at least  $n_1$  parallel edges connecting x to y. Therefore either  $\Gamma_1$  or  $\Gamma_2$  contains a Scharlemann cycle, by Corollary 2.6.7 and Lemma 2.6.8.

Type III. We shall use the same notation as in the corresponding case of the proof of Proposition 2.5.8. There are again three sub-cases.

(1) u, v, and x parallel. By Corollary 2.6.7, either  $\Gamma_2$  contains a Scharlemann cycle or  $|U| \le n_1/2$ ,  $|V| \le n_1/2$ . Then  $|W| \ge n_1$ . Therefore, by Claim 2.6.9,  $|W| = n_1$ . Since  $\Gamma_0$  has no boundary edges and no trivial loops, it follows that on the torus T, all the inner boundary components of  $P_1$  lie between some pair of boundary components of F. This is conclusion (iii) of the proposition.

(2) u, v parallel, antiparallel to x. By Claim 2.6.9,  $|W| \le n_1$ . Hence  $|U| + |V| \ge n_1$ , which implies that every vertex of  $\Gamma_1$  occurs as the label at x of some edge of  $\Gamma_2$  connecting x to either u or v. Then  $\Gamma_1$  contains a great x-cycle by Lemma 2.6.3, and consequently a Scharlemann cycle by Lemma 2.6.2.

(3) u, v antiparallel. We may assume without loss of generality that u is parallel to x. Let X be the set of labels around u of the edges of  $\Gamma_2$  connecting u to x. (See Figure 2.11.) By Lemma 2.6.8, we may assume that  $|V| < n_1$ , and hence that  $|U| + |W| > n_1$ . By Lemma 2.6.6, we may also assume that  $X \cap U = \emptyset$ . Therefore  $X \subset W$ . By the argument in case (1) above, we may assume that  $|W| < n_1$ . Then, since  $\Gamma_0$  has no boundary edges and no trivial loops, no boundary components of F occur between any two adjacent inner boundary components of  $P_1$  corresponding to the labels in W. Since  $X \subset W$ , the

> This content downloaded from 128.248.155.225 on Sun, 4 Jan 2015 21:37:06 PM All use subject to JSTOR Terms and Conditions

same is true for X, and hence (since  $\Gamma_0$  has no trivial loops) for U also. Since  $|U| + |W| > n_1$ , it follows that all the boundary components of F occur between some pair of inner boundary components of  $P_1$ , namely those corresponding to the two adjacent extremal labels of U and W. This is equivalent to conclusion (iii).

#### 2.7. Dehn surgery on cabled manifolds

In this section we complete the proof of the Cyclic Surgery Theorem. Let M be a compact, connected, irreducible 3-manifold with torus boundary, which is not Seifert-fibered. The following is an immediate consequence of Theorems 1.0.1, 2.0.1, 2.0.2, and 2.0.3, and Addendum 2.0.4.

COROLLARY 2.7.1. If  $\pi_1(M(r))$  and  $\pi_1(M(s))$  are cyclic groups, then either  $\Delta(r, s) \leq 1$  or M contains an essential torus which cobounds with  $\partial M$  a cable space in M.

Suppose, then, that  $M = M' \cup C_{p,q}$ , where  $C_{p,q}$  is a cable space of type (p, q) and  $\partial M'$  is incompressible in M'. Suppose that  $\pi_1(M(r_i))$  is cyclic, i = 1, 2. We must show that  $\Delta(r_1, r_2) \leq 1$ . Since  $\pi_1(M(r_i))$  is cyclic,  $\partial M'$ compresses in  $C_{p,q}(r_i)$ , i = 1, 2. By [Gr1, Lemma 7.2] (or [Gr-L, Lemma 3.1]), this implies that there are co-ordinates on  $\partial M$  such that either  $r_i = pq$  or  $r_i = (1 + k_i pq)/k_i$ . Since  $\Delta(pq_i(1 + kpq)/k) = 1$  for all k, it follows that either  $\Delta(r_1, r_2) \leq 1$  or  $r_i = (1 + k_i pq)/k_i$  for some integer  $k_i$ , i = 1, 2. Note that  $\Delta(r_1, r_2) = |k_1 - k_2|$ . Then  $C_{p,q}(r_i)$  is a solid torus ([Gr1, Lemma 7.2]), and hence  $M(r_i) \cong M'(r_i)$  for some slopes  $r'_1, r'_2$  on  $\partial M'$ . Moreover, there are co-ordinates on  $\partial M'$  such that  $r'_i = (1 + k_i pq)/k_i q^2$ , i = 1, 2 (see [Gr1, Corollary 7.3] or [Gr-L, Lemma 3.1]). Then  $\Delta(r_1', r_2') = |k_1 - k_2|q^2 =$  $\Delta(r_1, r_2)q^2$  which is greater than 1 if  $r_1 \neq r_2$ . Therefore applying Corollary 2.7.1 to M' we infer that  $M' = M'' \cup C_{p', q'}$ , say, where  $\partial M''$  is incompressible in M''. By the same argument that we originally applied to M, there are co-ordinates on  $\partial M'$  such that  $r_i' = (1 + k_i' p' q') / k_i'$ , i = 1, 2. But the argument in [Gr-L, p. 137] shows that this is incompatible with the previous co-ordinate expressions of the  $r'_{i}$ , unless q = 2 and  $|k_{1} - k_{2}| = \Delta(r_{1}, r_{2}) = 1$ .

This completes the proof of the Cyclic Surgery Theorem.

#### 2.8. Property P for symmetric knots

The purpose of this section is to give a proof of Corollary 7, which was stated in the introduction to the paper.

Let K be a non-trivial knot in  $S^3$  such that there exists a periodic automorphism h of  $S^3$ , not equal to the identity, satisfying h(K) = K. Our goal is to show that K has Property P. Replacing h by a suitable power, we may assume that h has prime order, say p. Let Fix(h) denote the fixed-point set of h.

We distinguish three cases.

(1) h orientation-reversing. In this case K is amphicheiral and the result follows from Corollary 4.

(2) *h* orientation-preserving,  $Fix(h) \cap K \neq \emptyset$ . By the Smith Conjecture [M-B],  $Fix(h) \neq K$ , so that  $Fix(h) \cap K \cong S^0$  and *h* is an involution whose restriction to *K* is orientation-reversing. Thus *K* is strongly invertible, and hence has Property P by [B-S].

(3) h orientation-preserving,  $Fix(h) \cap K = \emptyset$ . Let M be the complement of an h-invariant open tubular neighborhood of K, and let  $M^*$  be the quotient of M by h|M. Note that, here, the quotient map  $\partial M \to \partial M^*$  is a p-fold (cyclic) covering. For any slope r on  $\partial M$ , h|M extends to a periodic automorphism of M(r) (which may fix pointwise the core of the attached solid torus J). Let  $h_n$ denote the extension to M(1/n), where we are using the standard parametrization of slopes on  $\partial M$ . Note that if  $r^*$ ,  $s^*$  are slopes on  $\partial M^*$  which lift to slopes r, s on  $\partial M$ , then  $\Delta(r^*, s^*) = p\Delta(r, s)$  (consider the intersection between the pre-images of  $r^*$  and  $s^*$ , which consist of p copies of r and s respectively).

There are two sub-cases.

(A) Fix(h)  $\neq \emptyset$ . In this case Fix( $h_n$ ) = Fix(h). Let N be the complement in M of an h-invariant open tubular neighborhood of Fix(h), and let  $N^*$  be the quotient of N by h|N. Since the linking number of K and Fix(h) is non-zero, Nis irreducible, and hence so is  $N^*$ . With the obvious notation,  $h_n$  induces a free  $\mathbb{Z}_p$ -action on N(1/n) with quotient  $N^*(r_n^*)$  for some slope  $r_n^*$  on  $\partial M^*$ . (In fact, using the usual meridian-longitude co-ordinates for  $\partial M^*$ , we have  $r_n^* =$ 1/pn.) Now suppose that M(1/n) is simply-connected. Then, by the Generalized Smith Conjecture [M-B], Fix( $h_n$ ) is unknotted, so that N(1/n) is a homotopy solid torus. Hence  $N^*(r_n^*)$  is also a homotopy solid torus. Since  $M(1/0) \cong S^3$  is simply-connected, and since  $\Delta(r_n^*, r_0^*) = p\Delta(1/n, 1/0) > 1$  if  $n \neq 0$ , it follows from Theorem 2.4.4 applied to  $N^*$  that, if  $n \neq 0$ , then  $N^*$  is either a cable space or homeomorphic to  $\partial M^* \times I$ . Therefore N is either a cable space or homeomorphic to  $\partial M \times I$ , and hence (since Fix(h) is unknotted) K is a torus knot. Since non-trivial torus knots have Property P [Ms], we are done.

(B) Fix(h) =  $\emptyset$ . Suppose that M(1/n) is simply-connected.

If  $h_n$  fixes the core of J, then by the Generalized Smith Conjecture, M is a solid torus, and hence K is trivial.

If  $\operatorname{Fix}(h_n) = \emptyset$ , then  $h_n$  generates a free  $\mathbb{Z}_p$ -action on M(1/n), with quotient  $M^*(r_n^*)$  for some slope  $r_n^*$  on  $\partial M^*$ . Hence  $\pi_1(M^*(r_n^*))$  is cyclic. Since  $M(1/0) \cong S^3$ , the Cyclic Surgery Theorem implies that either

 $\Delta(r_n^*, r_0^*) \leq 1$  or  $M^*$  is Seifert-fibered. Since  $\Delta(r_n^*, r_0^*) = p\Delta(1/n, 1/0)$ , the first conclusion is impossible if  $n \neq 0$ . If the second conclusion holds, then M is also Seifert-fibered. Hence K is a torus knot, and therefore has Property P.

Institute for Advanced Study, Princeton, New Jersey University of Texas, Austin, Texas Courant Institute, New York University, New York University of Illinois, Chicago, Illinois

#### References

- [A-M] S. AKBULUT and J. MCCARTHY, Casson's invariant for oriented homology 3-spheres—an exposition, preprint.
- [B-S] S. BLEILER and M. SCHARLEMANN, A projective plane in  $\mathbb{R}^4$  with three critical points is standard; strongly invertible knots have Property P, preprint.
- [Bo] F. BONAHON, Cobordism of automorphisms of surfaces, Ann. Sc. Ec. Norm. Sup. 4<sup>e</sup> série 16 (1983), 237–270.
- [C] A. J. CASSON, An integer invariant of homology 3-spheres, to appear.
- [C-G] A. J. CASSON and C. MCA. GORDON, Reducing Heegaard splittings, to appear in Topology and its Applications.
- [C-G-L-S] M. CULLER, C. MCA. GORDON, J. LUECKE and P. B. SHALEN, Dehn surgery on knots, Bull. Amer. Math. Soc. 13 (1985), 43-45.
- [C-M] M. CULLER and J. W. MORGAN, Group actions on R-trees, to appear in Proc. London Math. Soc.
- [C-S1] M. CULLER and P. B. SHALEN, Varieties of group representations and splittings of 3-manifolds, Annals of Math. 117 (1983), 109–146.
- [C-S2] \_\_\_\_\_, Bounded, separating surfaces in knot manifolds, Invent. Math. 75 (1984), 537-545.
- [D-S] M. DOMERCUE and H. SHORT, Surfaces incompressibles dans les variétés obtenues par chirurgie à partir d'un noeud de S<sup>3</sup>, C. R. Acad. Sci. Paris 300 (1985), 669–672.
- [Ga] D. GABAI, Surgery on knots in solid tori, preprint.
- [Gn] F. GONZÁLEZ-ACUÑA, Dehn's construction on knots, Bol. Soc. Mat. Mex. 15 (1970), 58-79.
- [Gr1] C. McA. GORDON, Dehn surgery and satellite knots, Trans. Amer. Math. Soc. 275 (1983), 687–708.
- [Gr2] \_\_\_\_\_, On primitive sets of loops in the boundary of a handle body, to appear in Topology and its Appl.
- [Gr-L] C. MCA. GORDON and R. A. LITHERLAND, Incompressible planar surfaces in 3-manifolds, Topology and its Appl. 18 (1984), 121-144.
- [Hei] W. HEIL, Elementary surgery on Seifert fiber spaces, Yokohama Math. J. 22 (1974), 135-139.
- [Hem] J. HEMPEL, 3-Manifolds, Ann. of Math. Studies 86, Princeton Univ. Press, Princeton, New Jersey, 1976.
- [J] W. Jaco, Adding a 2-handle to a 3-manifold: An application to Property R, Proc. A.M.S. 92 (1984), 288-292.
- [Jb] N. JACOBSON, Lectures in Abstract Algebra III, Springer-Verlag, New York, 1980.
- [Jo] K. JOHANNSON, On surfaces in one-relator 3-manifolds, preprint.
- [K] T. KANETO, On simple loops on a solid torus of general genus, Proc. A.M.S. 86 (1982). 551-552.

300	M. CULLER, C. M. GORDON, J. LUECKE, P. B. SHALEN
[L]	R. A. LITHERLAND, Surgery on knots in solid tori, II, J. London Math. Soc. (2) 22 (1980), 559–569.
[M-B]	J. W. MORGAN and H. BASS (editors), The Smith Conjecture, Academic Press, New York 1984
[M-S1]	J. W. MORGAN and P. B. SHALEN, Valuations, trees, and degenerations of hyperbolic structures, I, Annals of Math. 120 (1984), 401–476.
[M-S2]	, Valuations, trees, and degenerations of hyperbolic structures, II, preprint.
[Ms]	L. MOSER, Elementary surgery along a torus knot, Pacific J. Math. 38 (1971), 737-745.
[N]	J. NIELSEN, Die Isomorphismen der allgemeinen, unendlichen Gruppe mit zwei Erzeugenden Math Ann 78 (1918) 385-397
[P]	J. H. PRZYTYCKI, Incompressibility of surfaces after Dehn surgery, Michigan Math. J. <b>30</b> (1983), 289–308.
[ <b>R</b> ]	D. ROLFSEN, Knots and Links, Math. Lect. Ser. 7, Publish or Perish, Berkeley, California, 1976.
[Sch1]	M. SCHARLEMANN, Outermost forks and a theorem of Jaco, Proc. Rochester Conf., AMS Contemporary Math. Series 44 (1985), 189–193.
[Sch2]	, Smooth spheres in $\mathbb{R}^4$ are standard, Invent. Math. 79 (1985), 125–141.
[Sco-W]	G. P. SCOTT and C. T. C. WALL, Topological methods in group theory; in <i>Homological Group Theory</i> , London Math. Soc. Lecture Notes 36, Cambridge Univ. Press, Cambridge 1979, pp. 137–203.
[Se]	J-P. SERRE (with H. Bass), Trees, Springer-Verlag, New York, 1980.
[Shf]	I. R. SHAFAREVICH, <i>Basic Algebraic Geometry</i> , Die Grundlehren der mathematischen Wissenschaften, Band 213, Springer-Verlag, New York, 1974.
[Sho]	H. SHORT, Some closed incompressible surfaces in knot complements which survive surgery, in <i>Low Dimensional Topology</i> , L. M. S. Lecture Note Series 95 (1985), 179–194.
[Wa1]	F. WALDHAUSEN, On irreducible manifolds which are sufficiently large, Annals of Math. 87 (1968), 56–88.
[Wa2]	, Some problems on 3-manifolds, Proc. Symp. Pure Math. 32, Part 2 (1978), 313-322.
[Wh]	W. WHITTEN, Knot complements and groups, preprint.
[Z]	H. ZIESCHANG, On simple systems of paths on complete pretzels, A.M.S. Transl. (2) 92 (1970), 127–137.
(Received May 28, 1985)	

This content downloaded from 128.248.155.225 on Sun, 4 Jan 2015 21:37:06 PM All use subject to JSTOR Terms and Conditions